T&N LIMITED AND THE OTHER CVA COMPANIES

TRUST DISTRIBUTION PROCEDURES FOR THE UK ASBESTOS TRUST

As amended by:

(i) the 2007 Amending Deed dated 1 May 2008

- (ii) the 2008 Amending Deed dated 21 May 2009
- (iii) the 2009 Amending Deed dated 10 December 2010
- (iv) (there was no 2010 Amending Deed)

(v) the 2011 (Fifth Anniversary) Amending Deed dated 19 May 2011 (product liability changes took effect on 17 March 2011 and all other changes to take effect on 11 October 2011)

(vi) the rounding of the revised Schedule 3 values to take effect on 11 October 2011 was corrected by the Trustees pursuant to clause 6.1 in July 2011

(vii) the 2012 Amending Deed dated 19 December 2012

(viii) the 2016(i) Amending Deed dated 23 June 2016

(ix) the 2016(ii) (Tenth Anniversary) Amending Deed dated 5 January 2017

Contents

1	Introduction	7
1.1	Definitions and Interpretation	7
1.2	Scope of the TDP	7
1.3	Core Objective	9
1.4	Payments	9
	T&N Funds	9
	Chester Street Funds	9
	Non-T&N Funds	9
	Agency Companies	10
	Additional Dividend	10
	EL Asbestos Insurance Fund	10
1.5	Trust Claimants and the Hercules Policy	11
1.6	Structure of the TDP	11
2	Establishment of Trust Claims	12
2.1	Advertising for Trust Claims	12
2.2	Time Limits on Trust Claims	12
2.3	Proofs of Claim	13
	General	13
	Asbestos Exposure by more than one party	13
	Deceased Injured Persons	13
	Disease Level V cases	14
2.4	Election by Trust Claimant for Expedited Review Process or Individual Review Process 14	
	Trust Claims eligible for Expedited Review	14
	Election	14
	Individual Review only	14
	Possible consequence of choosing Individual Review where Expedited Review available	15
2.5	Expedited Review Process	15

	Consideration of Trust Claims	15
	Valuation of Established Claims: general	17
	Valuation of Established Claims: clothing and neighbourhood claims [DELETED]	18
	Valuation of Established Claims: product liability claims	18
	Valuation of Established Claims: Cape Trust Claims	19
	Admission of Established Claims: T&N	19
	Admission of Established Claims: other Companies	19
	Admission of Established Claims: Agency Companies	20
	Admission of Established Claims: certain Agency Companies	21
	Rejection of Trust Claims	22
	Referral of the Trustees' determination to the Expert	22
	EL Scheme Trust Claims	22
2.6	Individual Review Process	23
	Introduction	23
	Consideration of Trust Claims	23
	Valuation of Established Claims: Pre-Administration Liquidated Claims	24
	Valuation of Established Claims: Indirect Asbestos Trust Claims	24
	Valuation of Established Claims: other categories of Trust Claim	24
	Valuation of Established Claims: product liability claims	26
	Valuation of Established Claims: Cape Trust Claims	26
	Admission of Established Claims: T&N	26
	Admission of Established Claims: other Companies	27
	Admission of Established Claims: Agency Companies	28
	Admission of Established Claims: certain Agency Companies	29
	Rejection of Trust Claims	30
	Referral of the Trustees' determination to the Expert	30
	EL Scheme Trust Claims	30
	EL Asbestos Insured Trust Claimants	31
2.7	Trust Claims must be for the most severe Disease Level	32
2.8	Subsequent Trust Claims	33

2.9 Persor	Special Provisions applicable to deceased Injured Persons and certain living Injured	33
2.10	CRU Trust Claims [DELETED]	34
2.11	Special Provisions applicable to joint liability Trust Claims	35
2.12	Trust Claims: co-operation of T&N and the Companies	35
2.13	Trustees' powers and procedures	36
2.14	Expert Determination of Disputes	37
3	Payments to Trust Claimants from the T&N Fund and the T&N Hercules Fund	39
3.1	Computation of the T&N Payment Percentage; Reviews	39
3.2	Suspension of Payments from the T&N Fund	39
3.3	Additional Dividend	40
	Value of Established Claims for the purpose of Additional Dividend	40
	Setting the level of Additional Dividend	40
	Suspension	41
3.4	Payments in respect of Established Claims from the T&N Fund	41
	First Payment	41
	Additional Dividend	41
	Costs Payments	41
	TDP Determines Payments	42
3.5	Payment Mechanics	43
3.6	Deductions from Payments and Repayments	43
	First payments in respect of Established Claims	43
	Repayment: no clawback except in cases of fraud	44
	Revising the value of the Established Claim	44
	Overpayments and Additional Dividend	44
3.7	The T&N Hercules Fund	45
4	Chester Street Trust Claims	47
4.1	Assertion of Chester Street Trust Claims	47
	Introduction	47
	Exceptions for Chester Street Trust Claims	47

	Specific application of Section 2	48
4.2 Cheste	Payments to Chester Street Trust Claimants from the Chester Street Fund and the er Street Hercules Fund	49
4.3	Computation of the Chester Street Payment Percentage; Reviews	49
4.4 Claims	Suspension of Payments from the Chester Street Fund and compromise of Trust 51	
4.5	Additional Dividend	52
4.6	Payments in respect of Established Claims	52
	First Payment	52
	Additional Dividend	53
	Payment(s) in accordance with Clause 4.6 and from the Chester Street Funds	53
4.7	Payment Mechanics	53
4.8	Deductions from Payments and Repayments	53
4.9	The Chester Street Hercules Fund	53
4.10 Clause	4.10 Payments to the T&N Hercules Fund and Chester Street Hercules Fund pursuant to Clauses 2.4.3(b) and 4.1.2(c) of the Trust Deed	
5	Payments in relation to Established Claims against the Non-T&N Funds	56
5.1	Non-T&N Fund Claims	56
5.2	Medical and legal costs	56
5.3	Payment Mechanics	57
5.4	Deductions from Payments and Repayments	57
5.5	The operation of Non-T&N Funds after the completion of the relevant CVA	58
Transf	ers under paragraphs 5.7.1 to 5.7.3 of the CVAs	58
Setting	g Payment Percentages for the Non-T&N Funds	59
Suspe	nsion of Payments from the Non-T&N Funds	59
Additic	onal Dividend	60
Payme	ents in respect of Established Claims from the Non-T&N Funds	60
6 costs d	Revision of Values, maximum figures for pain and suffering, Payment Percentages, cap, levels of Additional Dividend and medical costs.	61
6.1	General Provisions	61
6.2	Special provision for Values in Schedule 3	62

6.3 Percer	Special Provisions for the T&N Payment Percentage and the T&N Hercules Payment ntage	62
6.4 Hercul	Special Provisions for the Chester Street Payment Percentage and the Chester Street es Payment Percentage	t 63
6.5 comple	6.5 Special Provisions for the Payment Percentages for Non-T&N Funds after the completion of the relevant CVA	
7	UK Asbestos Trust Costs	64
7.1	Introduction	64
7.2	Non-T&N Fund Costs	64
7.3	Other UK Asbestos Trust Costs	65
	T&N Fund Costs and T&N Hercules Fund Costs	65
	Chester Street Fund Costs and Chester Street Hercules Fund Costs	65
	Costs not attributable to any one particular Fund	65
	Hercules Recovery Costs	66
	EL Asbestos Insurance Fund Costs	66
7.4	Remuneration Fund	66
8	Trustees' Powers of Amendment	67
9	Governing law and disputes	68
Sched	Schedule 1	
Sched	Schedule 2	
Sched	Schedule 3	

[Schedule 4 not shown]

SECTION ONE

1 Introduction

1.1 Definitions and Interpretation

- 1.1.1 In this TDP, unless the context otherwise requires or expressly provides:
 - (a) defined terms shall bear the meaning given to them in the Definitions and Interpretation Schedule; and
 - (b) the interpretation provisions set out in the Definitions and Interpretation Schedule shall apply to the interpretation of this TDP.
- 1.1.2 The term "litigation risk" is used in several of the amendments that have been made to this TDP since the Trust was established. The purpose of this Clause 1.1.2 is to explain the reason for this and the meaning of the term in this context. The consideration of Trust Claims by the Trustees is a two stage process on both Expedited Review and Individual Review. First, the Trustees determine whether a Trust Claim is valid and, if it is, the Trustees admit it as an Established Claim (Clauses 2.5.1 to 2.5.4A for Expedited Review and Clauses 2.6.1 to 2.6.4 for Individual Review). Secondly, if a Trust Claim is admitted as an Established Claim, the Trustees assess the value of the Established Claim (Clauses 2.5.5 to 2.5.7 for Expedited Review and Clauses 2.6.5 to 2.6.7D for Individual Review). While this approach is appropriate in most cases, it does not allow the Trustees adequately to reflect the uncertainty as to the validity of certain types of claim. If such claims were to go through a court process, there would be a significant risk that they would fail following (among other things) disclosure, cross examination and/or the resolution of a legal uncertainty. This is what is meant by "litigation risk" in this TDP. Discounts for litigation risk have been introduced with respect to certain types of claim on both Expedited Review and Individual Review with a view to ensuring that the Core Objective is met.

1.2 Scope of the TDP

- 1.2.1 This TDP sets out the procedures that must be followed by any Trust Claimant claiming a payment in respect of a Trust Claim from the Trust Funds. The categories of Trust Claims are:
 - (a) UK Asbestos PI Trust Claims;
 - (b) EL Scheme Trust Claims;
 - (c) Australian Asbestos PI Trust Claims;
 - (d) Cape Trust Claims;

- (e) Agency (CVA Asbestos) Trust Claims;
- (f) Indirect Asbestos Trust Claims;
- (g) Chester Street Trust Claims; and
- (h) EL Asbestos Insured Trust Claims.
- 1.2.2 This TDP also sets out the procedures the Trustees will follow in determining whether or not they should make a payment out of the Trust Funds and in determining the amount of the payment. No Trust Claimant has a right to receive a payment in respect of his Trust Claim unless it is made by the Trustees in accordance with the terms of this TDP.
- 1.2.3 A Trust Claim shall confer and be limited to conferring on the holder the right to receive such payment, if any, as is offered or payable by the Trustees in accordance with and subject to the terms of the Trust Deed and this TDP, and the right to pursue any remedies that the holder may have under the Trust Deed and this TDP against the Trustees in respect of his Trust Claim. For the avoidance of doubt, in the event that for any reason the Trustees have no funds enabling them to make any or any further payment(s) to the holder of any Trust Claim which should have been made in accordance with the terms of the Trust Deed and this TDP:
 - (a) the right of the holder of the Trust Claim remains strictly limited to asserting the Trust Claim; and
 - (b) the terms and provisions of the CVAs do not oblige, require or render liable T&N or any of the Companies in any circumstances to make any payment to the holder of any Trust Claim or the Trustees in respect of any sum due to the holder of a Trust Claim from the Trust.
- 1.2.4 The Trust Funds comprise the separate funds set out below (each one, a "Fund") created for the purpose of making payments in accordance with the terms of this TDP to each of the corresponding categories of Trust Claimant in so far as their Trust Claims are Established:
 - (a) In relation to T&N Trust Claimants:
 - (i) The T&N Fund; and
 - (ii) The T&N Hercules Fund,

together the "T&N Funds".

- (b) In relation to Chester Street Trust Claimants:
 - (i) The Chester Street Fund; and

(ii) The Chester Street Hercules Fund,

together the "Chester Street Funds".

- (c) In relation to Non-T&N Trust Claimants, separate Non-T&N Funds for each of the Companies.
- (d) In relation to EL Asbestos Insured Trust Claimants, the EL Asbestos Insurance Fund.

1.3 Core Objective

- 1.3.1 The Core Objective is to enable Trust Claimants with Established Claims to receive a payment (or payments) from the appropriate Fund or Funds which:
 - (a) reflects the value of the underlying CVA Asbestos Claim, the proceeds of which are assigned to the Trust by the Trust Claimant;
 - (b) is fair and proportionate, having regard to the interests of other Trust Claimants with similar Trust Claims claiming out of the same Fund or Funds; and
 - (c) is calculated and paid in an efficient and cost-effective manner, following an efficient and cost-effective assessment of the Trust Claim.

1.4 Payments

T&N Funds

1.4.1 The Trustees shall apply a Payment Percentage for the T&N Fund and, when recoveries are made under the Hercules Policy, a Payment Percentage for the T&N Hercules Fund, to all Established Claims in respect of T&N in accordance with Section 3 of this TDP.

Chester Street Funds

1.4.2 The Trustees shall apply a Payment Percentage for the Chester Street Fund and, when recoveries are made under the Hercules Policy, a Payment Percentage for the Chester Street Hercules Fund, to the Established Claims of Chester Street Trust Claimants in accordance with Section 4 of this TDP.

Non-T&N Funds

1.4.3 As to the Non-T&N Funds, until any transfer is made pursuant to paragraphs 5.7.1 to 5.7.3 of the CVAs of any cash remaining in the relevant Non-T&N Subreserve to the Trust, the Trustees shall submit Non-T&N Fund Claims in the CVA of the relevant Company. The dividend in respect of Non-T&N Fund Claims will be determined by the Supervisors of the CVA of the relevant Company, and paid by them to the Trustees, who will, subject to Clause 17 of the Trust Deed, pass it on in full to the Trust Claimant. After any transfer is made

pursuant to paragraphs 5.7.1 to 5.7.3 of the CVAs, the Trustees will make payments for Established Claims concerning Companies other than T&N as provided in Clause 5.5.

Agency Companies

- 1.4.4 Where the Trust Claim relates to an Agency Company, a Trust Claimant has two Trust Claims:
 - (a) A Trust Claim against the relevant Non-T&N Fund, giving rise to a Non-T&N Fund Claim, which the Trustees will assert in the CVA of the Company concerned prior to any transfer being made to the Trust pursuant to paragraphs 5.7.1 to 5.7.3 of the CVAs. After such a transfer has been made, the Trustees will pay Trust Claims from monies held in the Non-T&N Fund. This process is more fully explained in Clause 5.5 of this TDP.
 - (b) An Agency (CVA Asbestos) Trust Claim against the T&N Fund, arising from the agency relationship between the relevant Agency Company and T&N.

Where the Supervisors of the Agency Company are able to pay a 100% dividend on the Non-T&N Fund Claims, Trust Claimants will have no recourse to the T&N Fund or the T&N Hercules Fund for their Agency (CVA Asbestos) Trust Claims. Where the Supervisors of the Agency Company are unable to pay a 100% dividend on the relevant Non-T&N Fund Claims, Trust Claimants will be entitled to recover simultaneously from the Non-T&N Fund and the T&N Fund, save that they cannot recover in aggregate more than 100% of their Established Claims. Their Non-T&N Fund Claims and Agency (CVA Asbestos) Trust Claims shall abate rateably to the extent that they would otherwise recover more than 100%.

Additional Dividend

1.4.5 Any Additional Dividend shall be paid by the Trustees to Trust Claimants in accordance with Sections 3, 4, 5 and 6 of this TDP.

EL Asbestos Insurance Fund

1.4.6 As to the EL Asbestos Insurance Fund, Trust Claimants will be entitled to make a Trust Claim against the T&N Fund and/ or the Non-T&N Funds. They will also be entitled to make EL Asbestos Insured Trust Claims against the Trust. As to EL Asbestos Insured Trust Claims, the Trustees may, if they consider it appropriate in all the circumstances, make a claim against the relevant EL Insurer pursuant to the 1930 Act Rights assigned by the Trust Claimant to the Trust. The Trustees will pay to the Trust Claimant an amount equal to payments received from the relevant EL Insurer, less any EL Asbestos Insured Fund Costs and, pursuant to paragraph 28.2 of the CVAs, any EL Claims Handling Costs, that the Trustees consider are fairly attributable to that EL Asbestos Insured Trust Claim. These arrangements are explained fully in Clause 2.6.15.

1.5 Trust Claimants and the Hercules Policy

1.5.1 The Trust Claimants shall co-operate with the Trustees in the assertion of CVA Asbestos Claims against T&N or any of the other Companies for the purpose of generating Hercules Recoveries under the Hercules Policy.

1.6 Structure of the TDP

- 1.6.1 Section 2 of this TDP explains how to establish a Trust Claim against the T&N Funds and the Non-T&N Fund.
- 1.6.2 Section 3 of this TDP explains how payments will be made by the Trustees out of the T&N Funds.
- 1.6.3 Section 4 of this TDP explains how to establish a Chester Street Claim against the Chester Street Funds, and how payments will be made out of those Funds.
- 1.6.4 Section 5 of this TDP explains how the Trustees will in the first instance establish and lodge Non-T&N Fund Claims in the CVAs of relevant Companies, and how payments will be made in the implementation of the CVA in question. It also explains that in the circumstances set out in paragraphs 5.7.1 to 5.7.3 of the CVAs, cash remaining in the relevant Non-T&N Subreserve will be transferred to the Trust, and the Trustees will make payments from the relevant Non-T&N Fund for Established Claims in respect of Companies other than T&N.
- 1.6.5 Section 6 of this TDP explains how the Trustees will, on successive fifth anniversaries of the Effective Date and in some cases more frequently, review and if appropriate revise certain Values, maximum figures for pain and suffering, Payment Percentages, costs caps, levels of Additional Dividend and medical costs.
- 1.6.6 Section 7 of this TDP explains how the UK Asbestos Trust Costs will be paid.
- 1.6.7 Section 8 of this TDP explains in what circumstances the Trust Deed or TDP may be amended.
- 1.6.8 Section 9 of this TDP provides for the governing law of the TDP and the resolution of disputes arising out of or in connection with the Trust Deed, the TDP or the operations of the Trust.

SECTION TWO

2 Establishment of Trust Claims

2.1 Advertising for Trust Claims

The Trustees shall notify and/or advertise for Trust Claimants in the following manner:

- 2.1.1 As soon as reasonably practicable after the Effective Date, the Trustees shall write to all Trust Claimants of whom they are aware to give them notice of the establishment of the Trust and to invite them to submit a Proof of Claim in accordance with Clause 2.3 of this TDP.
- 2.1.2 The Trustees shall, as soon as reasonably practicable after the Effective Date, place advertisements in the Law Society Gazette, together with two national newspapers of their choice, giving notice that any person who considers that he has a Trust Claim should take steps to assert it and indicating how such person may obtain details of the Trust and this TDP and a Proof of Claim. Thereafter, the Trustees shall place advertisements in materially the same terms in the Law Society Gazette every year, and in two national newspapers of their choice on the fifth anniversary of the Effective Date, and every five years thereafter.
- 2.1.3 The Trustees shall not be obliged further to advertise for, or otherwise invite the submission of Trust Claims.

2.2 Time Limits on Trust Claims

- 2.2.1 The Trustees shall not admit as Established Claims Trust Claims relating to CVA Asbestos Claims which would have been statute-barred if made in proceedings commenced against T&N or the relevant Company when the Proof of Claim was lodged, unless:
 - (a) such proceedings were in fact issued before the expiry of the relevant limitation period for the CVA Asbestos Claim; or
 - (b) there is a standstill agreement in respect of the CVA Asbestos Claim with T&N or the relevant Company, and the limitation period for the CVA Asbestos Claim had not expired before the standstill agreement was entered.
- 2.2.2 Notwithstanding the above, the Trustees may in their absolute discretion admit a Trust Claim as an Established Claim if having regard to the provisions of section 33 of the Limitation Act 1980, section 19A of the Presumption and Limitation (Scotland) Act 1973, Article 50 of the Limitation (Northern Ireland) Order 1989, or where appropriate, provisions of other jurisdictions to similar effect, they consider it equitable to do so.
- 2.2.3 Should a Trust Claimant disagree with the Trustees' decision under Clause 2.2.2, the Trust Clamant shall within 28 days give the Trustees written notice that he wishes to refer the

matter to the Expert pursuant to Clause 2.14. The Expert's decision shall be final and binding.

2.3 Proofs of Claim

General

- 2.3.1 A Trust Claimant who wishes to make a Trust Claim for a payment out of the Trust Funds must lodge a Proof of Claim with the Trustees. Subject only to the rights of reimbursement of costs provided for in Clauses 3.4.4 to 3.4.7 and 5.2.1 to 5.2.4, he shall be responsible for the cost of completing and lodging the Proof of Claim, and providing such documentary or other evidence as the Trustees may require in accordance with Clause 2.13.1 below. To the extent practically possible, the Trustees will accept a single Proof of Claim from a Trust Claimant who wishes to submit Trust Claims in the EL Scheme Trust and this Trust.
- 2.3.2 Trust claimants are required to provide the information set out in the Proof of Claim appended at Schedule 4 to this TDP.

Asbestos Exposure by more than one party

2.3.3 Trust Claimants may make separate Trust Claims based on Asbestos Exposure by more than one party. All such Trust Claims must be made in one Proof of Claim. In the case of Agency (CVA Asbestos) Trust Claims, the Trust Claimant shall complete one Proof of Claim for the Non-T&N Trust Claim in respect of the relevant Company and the Agency (CVA Asbestos) Trust Claim in respect of T&N.

Deceased Injured Persons

- 2.3.4 Trust Claims in relation to deceased Injured Persons pursuant to the Statutes must be made in one Proof of Claim.
- 2.3.5 To avoid costs being incurred unnecessarily where Proofs of Claim are lodged in relation to deceased Injured Persons:
 - (a) where the payment in respect of a Trust Claim will not exceed £10,000, Clause 13 of the Trust Deed will apply, and the Proof of Claim may be lodged by the Personal Representative of the deceased Injured Person, or any person appearing to the Trustees to be a spouse or dependant of the deceased Injured Person, or any person appearing to the Trustees to be lawfully authorised to receive the payment on behalf of the deceased Trust Claimant;
 - (b) where the payment in respect of a Trust Claim will exceed £10,000, the Proof of Claim must be lodged by a Personal Representative, and where there is no Personal Representative, one must be appointed in order for the Trust Claim to be processed by the Trust.

Disease Level V cases

2.3.6 In Disease Level V cases, Proofs of Claim may be lodged. Payment in respect of Established Claims concerning Disease Level V will not be made unless there is a change in the law resulting in such claims being compensable. If such a change in the law occurs, the provisions of this TDP will apply.

2.4 Election by Trust Claimant for Expedited Review Process or Individual Review Process

Trust Claims eligible for Expedited Review

- 2.4.1 <u>The following Trust Claims are eligible for Expedited Review:</u>
 - (a) UK Asbestos PI Trust Claims;
 - (b) Cape Trust Claims notified to the Trustees before the first anniversary of the Effective Date;
 - (c) <u>subject to the written consent of the Trustees (to be granted or withheld at the</u> <u>Trustees' absolute discretion), Cape Trust Claims notified to the Trustees on or after</u> <u>the first anniversary of the Effective Date;</u>
 - (d) Agency (CVA Asbestos) Trust Claims; and
 - (e) EL Scheme Trust Claims.

are eligible for Expedited Review.

Election

2.4.2 A Trust Claimant whose Trust Claim is eligible for Expedited Review shall elect in his Proof of Claim whether he wishes the Trustees to adopt the Expedited Review Process described in Clause 2.5 or the Individual Review Process described in Clause 2.6 of this TDP. If he fails so to elect, he will be deemed to have elected to adopt the Expedited Review Process.

Individual Review only

- 2.4.3 The following claims can only proceed by way of Individual Review:
 - (a) Australian Asbestos PI Trust Claims,
 - (b) Cape Trust Claims notified to the Trustees <u>on or</u> after the first anniversary of the Effective Date <u>(save for where the Trustees have given written consent pursuant to</u> <u>Clause 2.4.1(c) above for a Cape Trust Claim to proceed by way of Expedited</u> <u>Review)</u>.
 - (c) EL Asbestos Insured Trust Claims,
 - (d) Indirect Asbestos Trust Claims, and
 - (e) Chester Street Trust Claims (as to which see Section 4).

- (f) together with neighbourhood and clothing claims as defined in Clause 2.5.4A where the Asbestos Exposure took place entirely before 1 October 1965. and
- (g) product liability claims as defined in Clause 2.5.6 where:
 - (i) <u>the product liability claim is for Disease Levels I or II and the Asbestos</u> Exposure took place entirely on or after 1 January 1976, or
 - (ii) the product liability claim is for Disease Levels III, IV or V and more than 20 percent of the period of Asbestos Exposure took place on or after 1 January 1976.

can only proceed by way of Individual Review.

Possible consequence of choosing Individual Review where Expedited Review available

2.4.4 If a Trust Claimant whose Trust Claim is eligible for Expedited Review chooses to proceed by way of Individual Review, he will not be entitled to recover Actual Legal Costs in accordance with Clauses 3.4.4 and 3.4.6, and Clauses 5.2.1, 5.2.3 and 5.5.8 (c), unless the Trustees or the Expert determine that the value of his Established Claim is at least 10% higher than the Expedited Review Value for the relevant Disease Level in the Expedited Review Process.

2.5 Expedited Review Process

Consideration of Trust Claims

- 2.5.1 Where the Trust Claimant elects to have his Trust Claim assessed under the Expedited Review Process, the Trustees shall consider whether the Trust Claim meets one or more of the Presumptive Medical and Exposure Criteria applicable to any of the Disease Level(s) set out in Schedule 1. The diagnosis of the medical criteria in Schedule 1 shall be undertaken by a Medical Expert, who must be an accredited respiratory specialist or a retired consultant in respiratory medicine in cases of Disease Levels I to IV, and must be a qualified doctor in cases of Disease Level V. Special provisions apply to EL Scheme Trust Claims, as to which see Clause 2.5.14 below.
- 2.5.2 The Trustees shall not be bound to accept the findings of the Medical Expert provided by the Trust Claimant and may obtain and rely on such further medical evidence as they think fit, including but not limited to the obtaining of a report from another Medical Expert in the event of uncertainty concerning the Injured Person's condition or where they consider the medical evidence provided to them by the Trust Claimant to be unsatisfactory.
- 2.5.3 The Trustees shall not be bound by the exposure evidence provided by the Trust Claimant and may obtain evidence from a consultant engineer or such other evidence as they see fit to assist them in determining the extent and relevance of any Asbestos Exposure alleged by a Trust Claimant.

- 2.5.4 If the Trustees determine that the Trust Claim meets all the relevant Presumptive Medical and Exposure Criteria for the relevant Disease Level they shall admit the Trust Claim as an Established Claim.
- 2.5.4A As to UK Asbestos PI Trust Claims brought by or in respect of (i) a person who suffered exposure to asbestos caused by the activities of T&N or any relevant Company where the asbestos dust was contained in an employee's clothes ("clothing claims"), or (ii) a person who resided or worked close to a manufacturing site of T&N or any relevant Company and suffered exposure to asbestos caused by the activities of the same ("neighbourhood claims"):
 - (a) For all Disease Levels, clothing and neighbourhood claims where the Asbestos Exposure in question occurred entirely before 1 October 1965 shall proceed by way of Individual Review, as made clear by Clause 2.4.3.
 - (b) For Disease Level I and II cases, where the Asbestos Exposure occurred partly before and partly on or after 1 October 1965, the Trustees shall consider the Trust Claims in the normal way provided by Clauses 2.5.1 to 2.5.4.
 - (c) For Disease Level II, III, IV and V cases, where the Asbestos <u>Exposure occurred partly before and partly on or after 1 October</u> <u>1965:</u>
 - (i) The Trustees shall apply percentage figures to each period of Asbestos Exposure reflecting the extent of Asbestos Exposure before 1 October 1965 and the extent of Asbestos Exposure on or after 1 October 1965.
 - (ii) If the Trustees consider it appropriate to admit the Trust Claim as an Established Claim they shall only do so in respect of the period of Asbestos Exposure that occurred on or after 1 October 1965, applying the percentage figure for that period to the Expedited Review Value for the relevant Disease Level.

2.5.4B As to product liability claims (as defined in Clause 2.5.6):

(a) For all Disease Levels, product liability claims where the Asbestos Exposure in question occurred entirely on or after 1 January 1976 shall proceed by way of Individual Review, in accordance with Clause 2.4.3.

- (b) For Disease Level I and II cases where the Asbestos Exposure occurred entirely before 1 January 1976, or partly before and partly on or after 1 January 1976, the Trustees shall consider the Trust Claims in the normal way provided by Clauses 2.5.1 to 2.5.4.
- (c) For Disease Level III, IV and V cases where the period of Asbestos Exposure began before 1 January 1976, if:
 - (i) it ended before 1 January 1976 or 20 per cent or less of the period of Asbestos Exposure took place on or after 1 January <u>1976, the Trustees shall consider the Trust Claims in the</u> normal way provided by Clauses 2.5.1 to 2.5.4.
 - (ii) more than 20 per cent of the period of Asbestos Exposure took place on or after 1 January 1976, the Trust Claims shall proceed by way of Individual Review, as stated in Clause 2.4.3.

Valuation of Established Claims: general

- 2.5.5 Where the Trustees determine that a UK Asbestos PI Trust Claim should be admitted as an Established Claim under the Expedited Review Process, they shall value the Established Claim as follows:
 - (a) In Disease Level I, and II and V-cases, the Trustees shall apply the Expedited Review Value for the relevant Disease Level set out in table 1 of Schedule 3. If it appears to the Trustees that the Asbestos Disease is attributable to any material extent to Asbestos exposure caused by the activities of any party other than T&N or the relevant Company, the Trustees shall apply the Apportionment Formula to the value of the Established Claim. If there is no sufficient evidence of material Asbestos Exposure by any other party, the Apportionment Formula shall not be applied.
 - (b) In Disease Level II cases, where the Trust Claimant was a smoker the Trustees shall apply a discount of 10% to the Expedited Review Value to reflect contributory negligence. If it appears to the Trustees that the Asbestos Disease is attributable to any material extent to Asbestos Exposure caused by the activities of any party other than T&N or the relevant Company, the Trustees shall apply the Apportionment Formula to the value of the Established Claim (as already reduced to reflect contributory negligence if applicable). If there is no sufficient evidence of material Asbestos Exposure by any other party, the Apportionment Formula shall not be applied.
 - (c) In Disease Level III and IV cases, the Trustees shall first consider the severity of the Asbestos Disease in respect of which the Trust Claim is made by reference to the

Disability Rating Scale set out in Schedule 2. They shall calculate the value of the Established Claim by applying the Expedited Review Value for mild, moderate or severe cases of the Asbestos Disease in table 1 of Schedule 3. If it appears to the Trustees that the Asbestos Disease is attributable to any material extent to Asbestos Exposure caused by the activities of any party other than T&N or the relevant Company, the Trustees shall apply the Apportionment Formula to the value of the Established Claim. If there is no sufficient evidence of material Asbestos Exposure by any other party, the Apportionment Formula shall not be applied.

- (d) In Disease Level V cases, the Trustees shall apply the Expedited Review Value for the relevant Disease Level set out in table 1 of Schedule 3. If it appears to the Trustees that the Asbestos Disease is attributable to any material extent to Asbestos Exposure by any party other than T&N or the relevant Company, the Trustees shall apply the Apportionment Formula to the value of the Established Claim. If there is no sufficient evidence of material Asbestos Exposure by any other party, the Apportionment Formula shall not be applied.
- (e) In cases where the Injured Person is deceased, the value of the Established Claim shall be set at the relevant Expedited Review Value for the relevant Disease Level set out in tables 2 and 3 of Schedule 3.
- (f) In all cases, if a Trust Claimant has received a payment from or on behalf of any other party in respect of his Asbestos Disease, he must disclose the amount of the payment to the Trustees. The Trustees may take the whole or part of the payment into account when valuing the Established Claim, if they consider it appropriate to do so.

Valuation of Established Claims: clothing and neighbourhood claims [DELETED]

Valuation of Established Claims: product liability claims

2.5.6 Where the Trustees determine that a UK Asbestos PI Trust Claim brought by or in respect of a person who suffered exposure to asbestos contained in product(s) manufactured by T&N or any relevant Company ("product liability claims") should be admitted as an Established Claim under the Expedited Review Process, the Trustees shall value the Established Claim as set out in Clause 2.5.5, and they shall then discount that value by one third 50% to reflect litigation risk.

Where the Trustees determine that a UK Asbestos PI Trust Claim brought by or in respect of (i) a person who suffered exposure to asbestos caused by the activities of T&N or any relevant Company, where the asbestos dust was contained in an employee's clothes ("clothing claims"), or (ii) a person who resided or worked close to a manufacturing site of T&N or any relevant Company and suffered exposure to asbestos caused by the activities of the same ("neighbourhood claims"), should be admitted as an Established Claim under the Expedited Review Process, the Trustees shall value the

Established Claim as set out in Clause 2.5.5, and they shall then discount that value by 20% to reflect litigation risk.

Valuation of Established Claims: Cape Trust Claims

2.5.7 Where the Trustees determine that a Cape Trust Claim should be admitted as an Established Claim under the Expedited Review Process, the Trustees shall value the Established Claim as set out in Clause 2.5.5, and they shall then discount that value by 70% 50% (provided that they may adjust that figure if they consider it appropriate to do so), to reflect litigation risk.

Admission of Established Claims: T&N

- 2.5.8 If the Established Claim is a Trust Claim in respect of T&N, the Trustees shall send the Trust Claimant an Expedited Review Admission Notice setting out in summary form the payment to be made to the Trust Claimant, showing:
 - (a) the value of the Established Claim which they have determined under the Expedited Review Process (in accordance with Clause 3.6.1 (b), there shall be no set-off of any amounts due in respect of a CRU Trust Claim against any amount due to the Trust Claimant);
 - (b) the net sum payable or estimated to be payable to the Trust Claimant after application of the then current T&N Payment Percentage or T&N Hercules Payment Percentage, out of the T&N Fund and/or the T&N Hercules Fund in accordance with Section 3; and
 - (c) any Costs Payments which the Trustees have determined should be paid pursuant to Clauses 3.4.4 to 3.4.7.

Admission of Established Claims: other Companies

- 2.5.9 If the Established Claim is a Trust Claim in respect of a Company other than T&N (except Agency Companies, as to which see Clause 2.5.10):
 - (a) The Trustees shall send an Expedited Review Admission Notice setting out in summary form the amount of the Established Claim showing the information required by Clause 2.5.8 (a) and the amount of any Costs Payments which the Trustees have determined should be paid pursuant to Clauses 5.2.1 to 5.2.4.
 - (b) The Trustees shall, subject to the completion of the expert process provided by Clauses 2.5.13 and 2.14, give notice of the Established Claim to the Supervisors of the relevant Company and submit a corresponding Non-T&N Fund Claim in the CVA for that Company. The Non-T&N Fund Claim shall be for a sum equal to the value of the Established Claim, including any Costs Payments that the Trustees determine should be paid pursuant to Clauses 5.2.1 to 5.2.4, with the intention that in

accordance with the terms of the CVA, the Non-T&N Fund Claim shall be Allowed for dividend on the value of the Established Claim in the relevant Company's CVA, together with the appropriate Costs Payments.

(c) Subject to clause 17 of the Trust Deed, the Trustees will pass on in full to the Trust Claimant all payments in respect of the Non-T&N Fund Claim that they receive from the relevant Company's Supervisors.

Admission of Established Claims: Agency Companies

- 2.5.10 If the Established Claim is a Trust Claim in respect of an Agency Company (except Newalls Insulation Company Limited, TBA Belting Limited, T&N Shelf Twenty-Six Limited, The Washington Chemical Company Limited, or JW Roberts Limited, as to which see Clause 2.5.11):
 - (a) The Trustees shall send an Expedited Review Admission Notice setting out in summary form the amount of the Established Claim showing the information required by Clause 2.5.8 (a) and the amount of any Costs Payments which the Trustees have determined should be paid pursuant to Clauses 5.2.1 to 5.2.4.
 - (b) The Trustees shall, subject to the completion of the expert process provided by Clauses 2.5.13 and 2.14, give notice of the Established Claim to the Supervisors of the relevant Agency Company and submit a corresponding Non-T&N Fund Claim in the CVA for that Agency Company. The Non-T&N Fund Claim shall be for a sum equal to the value of the Established Claim, plus:
 - where the relevant Agency Company's Supervisors are able to pay the Non-T&N Fund Claim in full, the full amount of any Costs Payments that the Trustees determine should be paid in accordance with Clauses 5.2.1 to 5.2.4;
 - (ii) where the relevant Agency Company's Supervisors are unable to pay the Non-T&N Fund Claim in full, 50% of any Costs Payments that the Trustees determine should be paid in accordance with Clauses 5.2.1 to 5.2.4,

in both cases with the intention that in accordance with the terms of the CVAs the Non-T&N Fund Claim shall be allowed for dividend on the value of the Established Claim in the relevant Agency Company's CVA, together with the appropriate Costs Payments.

(c) Subject to clause 17 of the Trust Deed, the Trustees will pass on in full to the Trust Claimant all payments in respect of the Non-T&N Fund Claim that they receive from the relevant Agency Company's Supervisors.

- (d) Where the relevant Agency Company's Supervisors are able to pay a 100% dividend on the relevant Non-T&N Fund Claim, the Trust Claimant will have no recourse to the T&N Fund or the T&N Hercules Fund in respect of his Agency (CVA Asbestos) Trust Claim.
- (e) For the avoidance of doubt, where the relevant Agency Company's Supervisors are unable to pay a 100% dividend on the relevant Non-T&N Fund Claim, the Trust Claimant will be entitled to recover simultaneously from the Non-T&N Fund and the T&N Fund, save that he cannot recover in aggregate more than 100% of his Established Claim. The Non-T&N Fund Claim and Agency (CVA Asbestos) Trust Claim shall abate rateably to the extent that he would otherwise recover more than 100%.

Admission of Established Claims: certain Agency Companies

- 2.5.11 If the Established Claim is a Trust Claim in respect of Newalls Insulation Company Limited, TBA Belting Limited, T&N Shelf Twenty-Six Limited, The Washington Chemical Company Limited, or JW Roberts Limited:
 - (a) The Trustees shall send an Expedited Review Admission Notice setting out in summary form the amount of the Established Claim showing the information required by Clause 2.5.8 (a) and the amounts of any Costs Payments which the Trustees have determined should be paid in accordance with Clauses 3.4.4 to 3.4.7.
 - (b) The Trustees shall, subject to the completion of the expert process provided by Clauses 2.5.13 and 2.14, give notice of the Established Claim to the Supervisors of the relevant Agency Company and submit a corresponding Non-T&N Fund Claim in the CVA for that Agency Company. The holder of each such Non-T&N Fund Claim shall receive a fixed sum payment of (as the case may be):
 - (i) £5 for Newalls Insulation Company Limited and TBA Belting Limited;
 - (ii) £60 for T&N Shelf Twenty-Six Limited;
 - (iii) £30 for The Washington Chemical Company Limited,;
 - (iv) £20 for JW Roberts Limited,

The Trustees shall not include any Costs Payments in the Non-T&N Fund Claim submitted in the CVA of the relevant Agency Company.

(c) Subject to Clause 17 of the Trust Deed, the Trustees will pass on in full to the Trust Claimant all payments in respect of the Non-T&N Fund Claim that they receive from the relevant Agency Company's Supervisors. (d) The Trust Claimant will be entitled to recover against the T&N Fund for his Agency (CVA Asbestos) Trust Claim, including any Costs Payments that the Trustees have determined should be paid.

Rejection of Trust Claims

2.5.12 If the Trustees determine that a Trust Claim should be rejected under the Expedited Review Process, they shall send the Trust Claimant an Expedited Review Rejection Notice setting out detailed reasons for their decision.

Referral of the Trustees' determination to the Expert

2.5.13 Any Trust Claimant who is dissatisfied with his Expedited Review Admission Notice or an Expedited Review Rejection Notice may elect to have his Trust Claim considered by the Expert pursuant to Clause 2.14 provided that he gives notice in writing to the Trustees of such election in the form set out in the Expedited Review Admission Notice or Expedited Review Rejection Notice (as the case may be) within 28 days of receipt of such notice. Upon giving notice of such election the Trust Claimant shall not be entitled to receive payment from the Trust Funds under the Expedited Review Process until the Expert has made his determination. In the case of Established Claims in respect of Companies other than T&N, the Trustees shall not submit any corresponding Non-T&N Fund Claim in the CVA of the relevant Company until the Expert has made his determination. If a Trust Claimant fails to give notice in accordance with this Clause he shall be bound by the determination of his Trust Claim by the Trustees under the Expedited Review Process and shall have no further right of appeal or review.

EL Scheme Trust Claims

- 2.5.14 As to EL Scheme Trust Claims assessed by Expedited Review:
 - If the underlying EL Scheme Claim was wholly admitted as an Established Claim by the EL Scheme Trustees:
 - (i) The Trustees shall treat the EL Scheme Trust Claim as an Established Claim against this Trust in the amount determined by the EL Scheme Trustees.
 - (ii) The Trustees shall deduct the total amount paid or payable by the EL Scheme Trustees to the EL Scheme Trust Claimant from the value of the Established Claim against this Trust.
 - (iii) The Trustees shall apply the relevant Payment Percentage(s) to the amount of the Established Claim less the sum deducted under sub-clause (a) (ii).
 - (iv) The Trustees shall send the Trust Claimant an Expedited Review Admission Notice setting out in summary form the information arising from sub-clauses

(a) (i) to (iii), and the net sum payable or estimated to be payable to the Trust Claimant from this Trust.

- The Trust Claimant shall not be entitled to any Costs Payments from this Trust.
- (b) If the underlying EL Scheme Claim was wholly rejected by the EL Scheme Trustees:
 - (i) The Trust Claimant shall have an EL Scheme Trust Claim against this Trust in the amount that the EL Scheme Trustees rejected.
 - (ii) The Trustees of this Trust shall be entitled to consider afresh whether the EL Scheme Trust Claim meets the Presumptive Medical and Exposure Criteria set out in Schedule 1 in accordance with Clauses 2.5.1 to 2.5.3. They shall either admit and value the EL Scheme Trust Claim as an Established Claim in accordance with Clauses 2.5.4 to 2.5.11, or reject it in accordance with Clause 2.5.12. The EL Scheme Trust Claimant may refer the Trustees' decision to the Expert in accordance with Clause 2.5.13.
- (c) If the underlying EL Scheme Claim was partly accepted and partly rejected by the EL Scheme Trustees:
 - (i) The Trustees shall apply the procedure set out in sub-clause (a) to that part of the EL Scheme Claim that the EL Scheme Trustees admitted.
 - (ii) The Trustees shall apply the procedure set out in sub-clause (b) to that part of the EL Scheme Claim that the EL Scheme Trustees rejected.

2.6 Individual Review Process

Introduction

- 2.6.1 Where a Trust Claimant elects or is required to have his Trust Claim assessed under the Individual Review Process the Trustees shall determine whether the Trust Claim should be admitted as an Established Claim in accordance with this Clause 2.6.
- 2.6.2 Special provisions apply in the case of EL Scheme Trust Claims and EL Asbestos Insured Trust Claims as set out in Clauses 2.6.14 and 2.6.15. In all other cases Clauses 2.6.3 to 2.6.13 inclusive apply.

Consideration of Trust Claims

2.6.3 Subject to Clause 2.6.2, the Trustees shall consider the Trust Claim by reference to the relevant common law and statutory provisions applicable to the jurisdiction in which the CVA

Asbestos Claim could have been brought, and shall have regard to the information given in the Proof of Claim, and all relevant circumstances, including but not limited to:

- (a) awards or settlement values of similar claims for the same Disease Level against T&N or any relevant Company or other comparable companies in the relevant jurisdiction;
- (b) any unsatisfied or part-satisfied actual award, judgment or settlement made or agreed pre-administration whether in respect of damages and/or legal costs and disbursements;
- (c) the strengths and weaknesses of the available evidence relating to the applicable Disease Level and medical and exposure criteria, including but not limited to the Presumptive Medical and Exposure Criteria set out in the table in Schedule 1 and the Disability Rating Scale set out in Schedule 2;
- (d) the age, symptoms and disability of the Injured Person;
- where appropriate, the extent to which the relevant disease is attributable to Asbestos Exposure for which T&N or the relevant Company, or any direct or indirect subsidiary of T&N, is not liable;
- (f) the extent to which injury and disability are attributable to some comorbid condition unrelated to Asbestos Disease.
- 2.6.4 The Trustees may obtain and rely on such legal, medical or other advice as they see fit for the purpose of the consideration of Trust Claims under Clause 2.6.3.

Valuation of Established Claims: Pre-Administration Liquidated Claims

2.6.5 The Trustees may value an Established Claim by reference to a Pre-Administration Liquidated Claim held by the Trust Claimant (save in the case of manifest error), but not so as to exceed the Maximum Value for the relevant Disease Level listed in Schedule 3.

Valuation of Established Claims: Indirect Asbestos Trust Claims

2.6.6 In the case of Indirect Asbestos Trust Claims, the Trustees shall value the Established Claim in such amount as they consider appropriate, taking into account all relevant facts and matters, but not exceeding the Maximum Value for the relevant Asbestos Disease suffered by the Injured Person.

Valuation of Established Claims: other categories of Trust Claim

2.6.7 In all other cases, where the Trustees determine that a Trust Claim should be admitted as an Established Claim under the Individual Review Process, they shall calculate the value of the Established Claim as follows:

- (a) In Disease Level I and, II and V cases, the value of the Established Claim shall be set at such amount as the Trustees shall determine but not exceeding the Maximum Value for the relevant Disease Level set out in table 1 of Schedule 3. If it appears to the Trustees that the Asbestos Disease is attributable to any material extent to Asbestos Exposure caused by the activities of any other party other than T&N or the relevant Company, the Trustees shall apply the Apportionment Formula to the value of the Established Claim. If there is no sufficient evidence of material Asbestos Exposure by any other party, the Apportionment Formula shall not be applied.
- (b) In Disease Level II cases, where the Trust Claimant was a smoker the Trustees may, if they consider it appropriate, apply a discount of up to 20% to the value that otherwise would have been fixed for Established Claims, to reflect contributory negligence. If it appears to the Trustees that the Asbestos Disease is attributable to any material extent to Asbestos Exposure caused by the activities of any party other than T&N or the relevant Company, the Trustees shall apply the Apportionment Formula to the value of the Established Claim (as already reduced to reflect contributory negligence if applicable). If there is no sufficient evidence of material Asbestos Exposure by any other party, the Apportionment Formula shall not be applied.
- (c) In Disease Level III and IV cases, the Trustees shall first consider the severity of the Asbestos Disease in respect of which the Trust Claim is made by reference to the Disability Rating Scale set out in Schedule 2. They shall calculate the value of the Established Claim by applying the Maximum Value for mild, moderate or severe cases of the Asbestos Disease in table 1 of Schedule 3. They shall set the value of the Established Claim at such amount as they shall determine, but not exceeding the Maximum Value for mild, moderate or severe cases of the Asbestos Disease in table 1 of Schedule 3. They shall set the value of the Established Claim at such amount as they shall determine, but not exceeding the Maximum Value for mild, moderate or severe cases of the Asbestos Disease in table 1 of Schedule 3. If it appears to the Trustees that the Asbestos Disease is attributable to any material extent to Asbestos Exposure caused by the activities of any party other than T&N or the relevant Company, the Trustees shall apply the Apportionment Formula to the value of the Established Claim. If there is no sufficient evidence of material Asbestos Exposure by any other party, the Apportionment Formula shall not be applied.
- (d) In Disease Level V cases, the value of the Established Claim shall be set at such amount as the Trustees shall determine but not exceeding the Maximum Value for Disease Level V set out in table 1 of Schedule 3. If it appears to the Trustees that the Asbestos Disease is attributable to any material extent to Asbestos Exposure by any party other than T&N or the relevant Company, the Trustees shall apply the Apportionment Formula to the value of the Established Claim. If there is no sufficient evidence of material Asbestos Exposure by any other party, the Apportionment Formula shall not be applied.

- (e) In cases where the Injured Person is deceased, the Trustees shall determine the value of the Established Claim at an amount which shall not exceed the Maximum Value for the relevant Disease Level set out in tables 2 and 3 of Schedule 3.
- (f) In all cases, if a Trust Claimant has received a payment from or on behalf of any other party in respect of his Asbestos Disease, he must disclose the amount of the payment to the Trustees. The Trustees may take the whole or part of the payment into account when valuing the Established Claim, if they consider it appropriate to do so.
- 2.6.7A In determining the value of Established Claims under Clause 2.6.7 above, the Trustees shall assume that the Injured Person will not develop any disease or suffer any deterioration in his condition that would entitle the Trust Claimant to make a subsequent claim under Clause 2.8 below. The Trustees shall only consider any such disease or deterioration if and when they receive a further Proof of Claim pursuant to Clause 2.8 below.
- 2.6.7B The Trustees may, however, take into account evidence of a chance that at some definite or indefinite time in the future the Injured Person will, as a result of the act or omission which gave rise to the Trust Claim, develop a disease or suffer a deterioration in his condition that would not entitle the Trust Claimant to make a subsequent claim under Clause 2.8 below. For example (without limitation) this means that the Trustees may take into account the risk of the Injured Person's percentage disability from Disease Levels III or IV worsening.

Valuation of Established Claims: product liability claims

2.6.7C Where the Trustees determine that a product liability claim (as defined in Clause 2.5.6 above) should be admitted as an Established Claim under the Individual Review Process, the Trustees shall value the Established Claim as set out in Clauses 2.6.5 to 2.6.7B (as appropriate). If the Trustees value the Established Claim pursuant to any of Clauses 2.6.6 to 2.6.7B, they shall then discount that value by 50% to reflect litigation risk. The discount shall not be applied where the Established Claim is valued pursuant to Clause 2.6.5.

Valuation of Established Claims: Cape Trust Claims

2.6.7D Where the Trustees determine that a Cape Trust Claim should be admitted as an Established Claim under the Individual Review Process, the Trustees shall value the Established Claim as set out in Clauses 2.6.5 to 2.6.7B (as appropriate). If the Trustees value the Established Claim pursuant to any of Clauses 2.6.6 to 2.6.7B, they may then in their discretion discount that value by 25%, 50% or 75% to reflect litigation risk. A discount shall not be applied where the Established Claim is valued pursuant to Clause 2.6.5.

Admission of Established Claims: T&N

2.6.8 If the Trustees determine that a Trust Claim in respect of T&N should be admitted under the Individual Review Process, they shall send the Trust Claimant an Individual Review Admission Notice setting out in summary form the payment to be made to the Trust Claimant, showing:

- the value of the Established Claim which they have determined under the Individual Review Process, broken down by heads of recovery;
- (b) any amounts due in respect of a CRU Trust Claim which may lawfully be set-off against any amount due to the Trust Claimant, in accordance with Clause 3.6.1 (b);
- (c) the net sum payable or estimated to be payable to the Trust Claimant after application of the then current T&N Payment Percentage/ T&N Hercules Payment Percentage in accordance with Section 3; and
- (d) any Costs Payments which the Trustees have determined should be paid, pursuant to Clauses 3.4.4 to 3.4.7.

Admission of Established Claims: other Companies

- 2.6.9 If the Established Claim is a Trust Claim in respect of a Company other than T&N (except Agency Companies, as to which see Clause 2.6.10):
 - (a) The Trustees shall send an Individual Review Admission Notice setting out in summary form the amount of the Established Claim showing the information required by Clause 2.6.8 (a) and (b), and the amount of any Costs Payments which the Trustees have determined should be paid pursuant to Clauses 5.2.1 to 5.2.4.
 - (b) The Trustees shall, subject to the completion of the expert process provided by Clauses 2.6.13 and 2.14, give notice of the Established Claim to the Supervisors of the relevant Company and submit a corresponding Non-T&N Fund Claim in the CVA for that Company. The Non-T&N Fund Claim shall be for a sum equal to the value of the Established Claim including any Costs Payments that the Trustees determine should be paid pursuant to Clauses 5.2.1 to 5.2.4, with the intention that in accordance with the terms of the CVA, the Non-T&N Fund Claim shall be Allowed for dividend on the value of the Established Claim in the relevant Company's CVA together with the appropriate Costs Payments.
 - (c) Subject to Clause 17 of the Trust Deed, the Trustees will pass on in full to the Trust Claimant all payments in respect of the Non-T&N Fund Claim that they receive from the relevant Company's Supervisors.

Admission of Established Claims: Agency Companies

- 2.6.10 If the Established Claim is a Trust Claim in respect of an Agency Company (except Newalls Insulation Company Limited, TBA Belting Limited, T&N Shelf Twenty-Six Limited, The Washington Chemical Company Limited or JW Roberts Limited, as to which see Clause 2.6.11):
 - (a) The Trustees shall send an Individual Review Admission Notice setting out in summary form the amount of the Established Claim showing the information required by Clause 2.6.8 (a) and (b), and the amount of any Costs Payments which the Trustees have determined should be paid pursuant to Clauses 5.2.1 to 5.2.4.
 - (b) The Trustees shall, subject to the completion of the expert process provided by Clauses 2.6.13 and 2.14, give notice of the Established Claim to the Supervisors of the relevant Agency Company and submit a corresponding Non-T&N Fund Claim in the CVA for that Agency Company. The Non-T&N Fund Claim shall be for a sum equal to the value of the Established Claim, plus:
 - where the relevant Agency Company's Supervisors are able to pay a dividend of 100% on the relevant Non-T&N Fund Claim, the full amount of any Costs Payments that the Trustees determine should be paid in accordance with Clauses 5.2.1 to 5.2.4;
 - (ii) where the relevant Agency Company's Supervisors are unable to pay a dividend of 100% on the Non-T&N Fund Claim, 50% of any Costs Payments that the Trustees determine should be paid in accordance with Clauses 5.2.1 to 5.2.4;

in both cases with the intention that in accordance with the terms of the CVA the Non-T&N Fund Claim shall be Allowed for dividend on the value of the Established Claim in the relevant Agency Company's CVA, together with the appropriate Costs Payments.

- (c) Subject to Clause 17 of the Trust Deed, the Trustees will pass on in full to the Trust Claimant all payments in respect of the Non-T&N Fund Claim that they receive from the relevant Agency Company's Supervisors.
- (d) Where the relevant Agency Company's Supervisors are able to pay a dividend of 100% on the relevant Non-T&N Fund Claim, the Trust Claimant will have no recourse to the T&N Fund or the T&N Hercules Fund in respect of his Agency (CVA Asbestos) Trust Claim.
- (e) For the avoidance of doubt, where the relevant Agency Company's Supervisors are unable to pay a 100% dividend on the relevant Non-T&N Fund Claim, the Trust

Claimant will be entitled to recover simultaneously from the Non-T&N Fund and the T&N Fund, save that he cannot recover in aggregate more than 100% of his Established Claim. The Non-T&N Fund Claim and Agency (CVA Asbestos) Claim shall abate rateably to the extent that he would otherwise recover more than 100%.

Admission of Established Claims: certain Agency Companies

- 2.6.11 If the Established Claim is a Trust Claim in respect of Newalls Insulation Company Limited, TBA Belting Limited, T&N Shelf Twenty-Six Limited, The Washington Chemical Company Limited or JW Roberts Limited:
 - (a) The Trustees shall send an Individual Review Admission Notice setting out in summary form the amount of the Established Claim showing the information required by Clause 2.6.8 (a) and (b), and the amount of any Costs Payments which the Trustees have determined should be paid in accordance with Clauses 3.4.4 to 3.4.7.
 - (b) The Trustees shall, subject to the completion of the expert process provided by Clauses 2.6.13 and 2.14, give notice of the Established Claim to the Supervisors of the relevant Agency Company and submit a corresponding Non-T&N Fund Claim in the CVA for that Agency Company. The holder of each such Non-T&N Fund Claim shall receive a fixed sum payment of (as the case may be):
 - (i) £5 for Newalls Insulation Company Limited and TBA Belting Limited;
 - (ii) £60 for T&N Shelf Twenty-Six Limited;
 - (iii) £30 for The Washington Chemical Company Limited,;

(iv) £20 for JW Roberts Limited,

The Trustees shall not include any Costs Payments in the Non-T&N Fund Claim submitted in the CVA of the relevant Agency Company.

- (c) Subject to Clause 17 of the Trust Deed, the Trustees will pass on in full to the Trust Claimant all payments in respect of the Non-T&N Fund Claim that they receive from the relevant Agency Company's Supervisors.
- (d) The Trust Claimant will be entitled to recover against the T&N Fund for his Agency (CVA Asbestos) Trust Claim, including any Costs Payments that the Trustees have determined should be paid.

Rejection of Trust Claims

2.6.12 If the Trustees determine that a Trust Claim shall be rejected under the Individual Review Process, they shall send the Trust Claimant an Individual Review Rejection Notice setting out detailed reasons for their decision.

Referral of the Trustees' determination to the Expert

2.6.13 Any Trust Claimant who is dissatisfied with an Individual Review Admission Notice or an Individual Review Rejection Notice may elect to have his Trust Claim determined by the Expert under the provisions of Clause 2.14 provided that he gives notice in writing to the Trustees of such election in the form set out in the Individual Review Admission Notice or Individual Review Rejection Notice (as the case may be) within 28 days of such receipt. Upon giving notice of such election the Trust Claimant shall lose any right to receive payment from the Trust Funds in accordance with the Individual Review Process, and in the case of Established Claims in respect of Companies other than T&N, the Trustees shall not submit any corresponding Non-T&N Fund Claim in the CVA of the relevant Company, until the Expert has made his determination. If the Trust Claimant fails to give notice in accordance with this Clause he shall be bound by the determination of his Trust Claim by the Trustees under the Individual Review Process and shall have no further right of appeal or review.

EL Scheme Trust Claims

- 2.6.14 As to EL Scheme Trust Claims assessed by Individual Review:
 - (a) If the underlying EL Scheme Claim was wholly admitted as an Established Claim by the EL Scheme Trustees:
 - The Trustees shall treat the EL Scheme Trust Claim as an Established Claim against this Trust in the amount determined by the EL Scheme Trustees, subject to sub-clause (ii).
 - (ii) The Trustees may value the Established Claim by reference to a Pre-Administration Liquidated Claim held by the EL Scheme Trust Claimant (save in the case of manifest error), but not so as to exceed the Maximum Value for the relevant Disease Level listed in Schedule 3.
 - (iii) The Trustees shall deduct the total amount paid or payable by the EL Scheme Trustees to the EL Scheme Trust Claimant from the value of the Established Claim against this Trust.
 - (iv) The Trustees shall apply the relevant Payment Percentage(s) to the amount of the Established Claim less the sum deducted under sub-clause (iii).

- (v) The Trustees shall send the Trust Claimant an Individual Review Admission Notice setting out in summary form the information arising from sub-clauses
 (i) to (iv), and the net sum payable or estimated to be payable to the Trust Claimant from this Trust.
- (vi) The Trust Claimant shall not be entitled to any Costs Payments from this Trust.
- (b) If the underlying EL Scheme Claim was wholly rejected by the EL Scheme Trustees:
 - (i) The Trust Claimant shall have an EL Scheme Trust Claim in the amount that the EL Scheme Trustees rejected.
 - (ii) The Trustees of this Trust shall be entitled to consider the EL Scheme Trust Claim afresh in accordance with Clauses 2.6.3 to 2.6.5. They shall either admit and value the EL Scheme Trust Claim as an Established Claim in accordance with Clauses 2.6.6 to 2.6.11, or reject it in accordance with Clause 2.6.12. The EL Scheme Trust Claimant may refer the Trustees' decision to the Expert in accordance with Clause 2.6.13.
- (c) If the underlying EL Scheme Claim was partly accepted and partly rejected by the EL Scheme Trustees:
 - (i) The Trustees shall apply the procedure set out in sub-clause (a) to that part of the EL Scheme Claim that the EL Scheme Trustees admitted.
 - (ii) The Trustees shall apply the procedure set out in sub-clause (b) to that part of the EL Scheme Claim that the EL Scheme Trustees rejected.

EL Asbestos Insured Trust Claimants

- 2.6.15 As to EL Asbestos Insured Trust Claimants:
 - (a) EL Asbestos Insured Trust Claimants will be entitled to make a Trust Claim against the T&N Fund and/ or the Non-T&N Funds. They will also be entitled to make EL Asbestos Insured Trust Claims against the EL Asbestos Insurance Fund.
 - (b) The Trustees shall assess Trust Claims against the T&N Fund and/ or the Non-T&N Funds in accordance with Clauses 2.6.3 to 2.6.5. They may admit and value such Trust Claims as Established Claims in accordance with Clauses 2.6.6 to 2.6.11 (subject to Clause 2.6.15 (c)), or reject them in accordance with Clause 2.6.12. Trust Claimants may refer the Trustees' decision to the Expert in accordance with Clause 2.6.13.

- (c) The Trustees shall deduct from the amount of any Established Claim against the T&N Fund or Non-T&N Fund a sum equal to the amount paid or payable to the Trust Claimant under sub-clause (e) in respect of the EL Asbestos Insured Trust Claim held by the Trust Claimant.
- (d) As to the EL Asbestos Insured Trust Claim, the Trustees may, if they consider it appropriate in all the circumstances, make a claim against the relevant EL Insurer pursuant to the 1930 Act Rights assigned by the Trust Claimant to the Trust.
- (e) The Trustees will pay to the Trust Claimant an amount equal to payments received from the relevant EL Insurer, less (i) any EL Asbestos Insured Fund Costs that the Trustees consider are fairly attributable to that EL Asbestos Insured Trust Claim, and (ii) any EL Claims Handling Costs payable in relation to that EL Asbestos Insured Trust Claim pursuant to paragraph 28.2 of the CVAs.
- (f) Notwithstanding the preceding sub-clauses, the Trustees shall have an absolute discretion to make payments to Trust Claimants in respect of their Established Claims against the T&N Fund and/ or the Non-T&N Funds prior to receiving any monies from the relevant EL Insurer. In these circumstances:
 - the Trustees shall not be required to make a payment under sub-clause (e) save to the extent that the sum received from the relevant EL Insurer which is payable to the Trust Claimant exceeds any sums which have already been paid to the Trust Claimant out of the T&N Fund and/ or the Non-T&N Fund concerned;
 - (ii) the Trustees will be entitled to reimburse the T&N Fund and/ or the relevant Non-T&N Fund (as they consider appropriate) for the value of payment(s) made out of such Funds from any monies which may subsequently be received by them from the relevant EL Insurer in respect of the EL Asbestos Insured Trust Claim; and
 - (iii) in reimbursing the T&N Fund and/ or the relevant Non-T&N Fund, the Trustees shall, in so far as reasonably practicable, restore the position to what it would have been if sub-clause (c) had been strictly applied.

2.7 Trust Claims must be for the most severe Disease Level

2.7.1 A Trust Claimant who makes a Trust Claim must claim for the most severe Disease Level for which he considers he qualifies at the time of lodging his Proof of Claim with the Trustees. For this purpose Disease Level I shall be the most severe and Disease Level V shall be the least severe. Irrespective of the Disease Level alleged on the Proof of Claim, each Trust Claim shall be treated as if it were a Trust Claim for the most severe Disease Level for which

the Trust Claimant is determined by the Trustees to qualify at the time of lodging the Proof of Claim with the Trustees.

2.8 Subsequent Trust Claims

- 2.8.1 Where relevant, a Trust Claimant who lodges a Proof of Claim in respect of a Trust Claim involving a non-malignant Asbestos Disease (Disease Levels III to V) may subsequently lodge a further Proof of Claim in respect of a Trust Claim involving a malignant Asbestos Disease (Disease Levels I or II) provided that the Trust Claimant did not know of the diagnosis of the malignant Asbestos Disease until after the Proof of Claim in respect of the non-malignant Asbestos Disease was lodged.
- 2.8.2 A Trust Claimant who lodges a Proof of Claim in respect of a Trust Claim involving Disease Levels IV or V may subsequently lodge a further Proof of Claim in respect of a Trust Claim involving asbestosis (Disease Level III) provided that the Trust Claimant did not know of the diagnosis of asbestosis until after the Proof of Claim in respect of Disease Levels IV or V was lodged.
- 2.8.3 Any subsequent Trust Claims made under this Clause must proceed by the Individual Review Process.

2.9 Special Provisions applicable to deceased Injured Persons and certain living Injured Persons

- 2.9.1 Subject to Clause 2.9.3 below, where a Trust Claim is lodged in respect of a living Injured Person, the first table at Schedule 3 to the TDP shall apply in determining the value of the Established Claim.
- 2.9.2 Where a Trust Claim is lodged in relation to a deceased Injured Person, the second and third tables at Schedule 3 to the TDP shall apply in determining the value of the Established Claim.
- 2.9.3 Where a Trust Claim is lodged by a living Injured Person in respect of Disease Levels I and II, who confirms in writing that he has no Dependants or Relatives <u>or any person who is akin to a Dependant or Relative under equivalent legislation</u>, or his Dependants or Relatives <u>or any such persons</u> confirm in writing that they will not bring any Claims pursuant to the Statutes on the death of the Injured Person, the first table at Schedule 3 to this TDP shall apply in determining the value of the Established Claim. Where there is no such confirmation, the amount of the pain and suffering element of the Trust Claim will be valued (subject to <u>fixed sum valuations of £70,000 in Disease Level I cases and £60,000 in Disease Level I cases assessed by Expedited Review and subject to</u> a maximum of £70,000 in Disease Level I cases and a maximum of £60,000 in Disease Level II cases <u>assessed by Individual Review</u>, provided that such valuations shall be reduced by applying the Apportionment Formula if the Apportionment Formula would otherwise apply to such living Injured Person's Established Claim pursuant to Clause 2.5.5(b) or Clause 2.6.7(b)), and the remainder of the Trust Claim

(the value of which will be determined in accordance with the second and third tables at Schedule 3 to the TDP and which shall be similarly reduced by applying the Apportionment Formula if it would otherwise be applicable pursuant to Clause 2.5.5(b) or Clause 2.6.7(b)) may be claimed by the individual who, on the death of the Injured Person, lodges a Trust Claim pursuant to Clause 2.3.5 of the TDP.

- 2.9.4 The aggregate of the values attributed to the Established Claim in respect of the Asbestos Disease of any one Injured Person will not exceed:
 - the relevant Expedited Review Value set out in Schedule 3 to this TDP, where the Trust Claims proceed by the Expedited Review Process; or
 - (b) the relevant Maximum Value set out in Schedule 3 to this TDP, where the Trust Claims proceed by the Individual Review Process,

so that where the living Injured Person or his Dependants or Relatives <u>or persons akin to his</u> <u>Dependants or Relatives under any equivalent legislation</u> provide the confirmation referred to in Clause 2.9.3, the Trust Claim is admitted by the Trustees, and the Established Claim is valued according to the second or third table of Schedule 3 to the TDP, no additional Trust Claims will be accepted in respect of that Injured Person's Asbestos Disease.

2.9.5 [DELETED] Where the Trustee determine that a Trust Claim made by a deceased Injured Person's Personal Representative should be admitted as an Established Claim, and the Personal Representative requests that the Trustees apportion the value of the Established Claim between the Injured Person's Dependents or Relatives and the Injured Person's estate, the Trustees will set out this apportionment in the Expedited Review Admission Notice or the Individual Review Admission Notice. Where a Trust Claimant is dissatisfied with this apportionment, Clauses 2.5.13 and 2.6.13 apply.

2.10 CRU Trust Claims [DELETED]

- 2.10.1 As to Trust Claims admitted as Established Claims to be paid out of the T&N Fund and the T&N Hercules Fund, the Trustees shall admit any CRU Trust Claim which is made and which in their absolute discretion they consider properly arises out of the relevant Established Claim. Any such CRU Trust Claim shall be admitted as an Established Claim in the amount determined by the Trustees but subject to the T&N Payment Percentage.
- 2.10.2 As to Trust Claims admitted as Established Claims in respect of Companies other than T&N, the CRU may make a CRU Trust Claim arising out of the Established Claim in relation to the relevant Company in the amount determined by the Trustees but subject to the Payment Percentage applicable to the relevant Non-T&N Fund. If the Company is an Agency Company, the CRU will also have an Agency (CVA Asbestos) Trust Claim in relation to T&N but subject to the T&N Payment Percentage. In such cases, the CRU will be entitled to recover simultaneously from the Non-T&N Fund and the T&N Fund, save that it cannot

recover in aggregate more than 100% of its Established Claim. The Non-T&N Fund Claim and Agency (CVA Asbestos) Trust Claim shall abate rateably to the extent that it would otherwise recover more than 100%.

2.10.3 Subject to Clauses 2.10.1 and 2.10.2, the CRU shall be treated as a Trust Claimant. CRU Trust Claims must all be made under the Individual Review Process described in Clause 2.6.

2.11 Special Provisions applicable to joint liability Trust Claims

2.11.1 If a Trust Claimant submits a Trust Claim to this Trust, and, prior to the Trust Claim being determined by the Trustees, a Trust Claimant has received a payment from or on behalf of any other party in respect of his Asbestos Disease caused by the same Asbestos Exposure, he must disclose the amount of the payment to the Trustees, and the amount allowable as an Established Claim shall be reduced by the amount of the payment.

2.12 Trust Claims: co-operation of T&N and the Companies

- 2.12.1 Pursuant to Clause 8 of the Trust Deed, T&N and the Companies shall co-operate with the Trustees in the Trustees' consideration of Trust Claims in the following manner:
 - (a) T&N and the Companies shall promptly provide the Trustees with such information and documents as the Trustees may reasonably request. For the avoidance of doubt, T&N and the Companies shall not in any circumstances provide or be required to provide any legally privileged information or documents to the Trustees, or take any step which constitutes or gives rise to a breach of the Hercules Policy. The Trustees will return and refrain from using or relying upon any legally privileged document which they are informed post-disclosure was disclosed by mistake.
 - (b) The Trustees shall reimburse any costs, fees and expenses (other than salaries, management time and overhead costs) reasonably incurred by T&N or any of the Companies in providing information or documents pursuant to a request made under Clause 2.12.1 (a). The Trustees shall not, however, reimburse T&N or any of the Companies for any costs incurred in maintaining their books, records, documents and control systems. T&N and the Companies shall not have any obligation to maintain or incur the cost of maintaining any microfiche reader, photocopier, or personnel to retrieve documents.
- 2.12.2 On or after the Hercules Policy Expiry Date, at the joint cost of the Trustees and the US Asbestos Trust, T&N may (but shall not be obliged to) transfer and assign, or cause to be transferred and assigned, jointly to the Trustees and the US Asbestos Trust the books, records, documents and control systems of T&N or any of the Companies that pertain directly to Asbestos Personal Injury Claims that have been asserted against T&N or any Company. From the date of any such transfer or assignment, neither T&N nor any of the Companies will be liable or responsible for the maintenance or cost of maintaining the books, records,

documents and control systems so transferred or assigned. If so requested by the EL Scheme Trustees, T&N and the Companies shall request that the Trustees provide such information and documents as the EL Scheme Trustees may reasonably request, and the Trustees shall promptly comply with such request.

2.12.3 If any books, records, documents and control systems are transferred or assigned to the Trustees and the US Asbestos Trust pursuant to clause 2.12.2, T&N and the Companies shall, at their own cost, be entitled to access to such books, records, documents and control systems, and the Trustees and the US Asbestos Trust shall (to the extent that it is within their power) promptly allow T&N or the Companies such access.

2.13 Trustees' powers and procedures

- 2.13.1 The Trust Claimant shall co-operate with the Trustees in the Trustees' consideration of Trust Claims. In particular, the Trust Claimant shall provide the Trustees with details of any matter required to be specified in the Proof of Claim, including further medical evidence or exposure evidence, if the Trustees request it.
- 2.13.2 If a Trust Claimant or any legal advisers identified by him in the Proof of Claim do not respond to a reasonable request for information by the Trustees under Clause 2.13.1, the Trustees shall be entitled to serve a notice giving the Trust Claimant or his legal advisers a further period of 28 days to respond, failing which the Trust Claim will be rejected.
- 2.13.3 A Trust Claim may be admitted by the Trustees either for the whole amount claimed by the Trust Claimant or for part of that amount, subject to the Expedited Review Values and Maximum Values provided by Schedule 3.
- 2.13.4 A Proof of Claim may at any time, by agreement between a Trust Claimant and the Trustees, be withdrawn or varied.
- 2.13.5 The Trustees may, if they consider it to be in the best interests of Trust Claimants generally, enter into any compromise or arrangement with any Trust Claimant or any person claiming to be a Trust Claimant.
- 2.13.6 Acceptance of a CVA Asbestos Claim for the purposes of voting at any CVA Meeting shall not bind the Trustees to accept either the existence or the amount of the corresponding Trust Claim.
- 2.13.7 The Trustees shall keep all documentation submitted to them for a period of 6 years after receipt, although they may if they so decide keep it in electronic form only.

2.14 Expert Determination of Disputes

- 2.14.1 Where a Trust Claimant elects to have his Trust Claim referred to the Expert under Clauses 2.5.13 or 2.6.13, or refers the Trustees' decision under Clause 2.2.2 to the Expert, this Clause shall apply.
- 2.14.2 The Expert shall act as an expert and not as an arbitrator.
- 2.14.3 The Expert shall assess the Trust Claim referred to him de novo, rather than limiting himself to a review of the Trustees' determination.
- 2.14.3A For the avoidance of doubt, the Expert's assessment of the Trust Claim shall be subject to the same provisions of this TDP as the Trustees' assessment (and, for this purpose, the word "Trustees" in those provisions shall be read as "Expert"), such that:
 - (a) <u>he shall consider Trust Claims referred to him under Clause 2.2.3 by reference to</u> <u>Clauses 2.2.1 and 2.2.2;</u>
 - (b) <u>he shall consider Trust Claims referred to him from the Expedited Review process by</u> reference to Clauses 2.5.1 to 2.5.4A and he shall value Established Claims by reference to Clauses 2.5.5 to 2.5.7, 2.9 and 2.11, as appropriate (including where any of those Clauses is applied by virtue of other Clauses including but not limited to Clause 2.5.14); and
 - (c) <u>he shall consider Trust Claims referred to him from the Individual Review process by</u> reference to Clauses 2.6.1 to 2.6.4 and he shall value Established Claims by reference to Clauses 2.6.5 to 2.6.7D, 2.6.14(a)(ii), 2.9 and 2.11, as appropriate (including where any of those Clauses is applied by virtue of other Clauses including but not limited to Clauses 2.6.14, 2.6.15 and 4.1).
- 2.14.4 Each party to the dispute shall, within 28 days of notification by the Trust Claimant to the Trustees of his wish to refer the matter to the Expert, provide the Expert with his written representations and any evidence on which he wishes to rely regarding the matter. The Expert shall have the power to extend the period in which such representations can be made.
- 2.14.5 Where the Trustees assessed the Trust Claim by way of Expedited Review, the Expert shall consider, in particular, whether the Trust Claimant satisfied the Presumptive Medical and Exposure Criteria set out in Schedule 1 on the basis of the information submitted by him to the Trustees, and any written representations or evidence provided to the Expert under Clause 2.14.4. Where the Trustees assessed the Trust Claim by way of Individual Review, the Expert shall consider, in particular, whether the Trust Claimant established his Trust Claim on the basis of the information submitted by him to the Trustees, and any written representations or evidence for the Expert Claim on the basis of the information submitted by him to the Trustees, and any written representations or evidence provided to the Expert under Clause 2.14.4. In both cases, the Expert may rely on evidence supplied by one party in the absence of evidence to the contrary

from any other party, provided always that such matters or evidence are relevant to the criteria or factors, as appropriate, which are laid down in Schedule 1 or Clause 2.6.3 as the case may be.

- 2.14.6 Each party to the dispute shall respond promptly (and at their own expense) to all reasonable requests for information and other assistance that the Expert makes in connection with his determination.
- 2.14.7 If a Trust Claimant or any legal advisers identified by him in the Proof of Claim do not respond to a reasonable request for information by the Expert under Clause 2.14.6, the Trustees shall be entitled to serve a notice giving the Trust Claimant a further period of 28 days to respond, failing which the Trust Claim will be rejected.
- 2.14.8 The decision of the Expert shall be final and binding on the parties and there shall be no right of review or appeal.
- 2.14.9 The Expert shall not be entitled to make a determination of a Trust Claim in excess of:
 - in Expedited Review cases, the Expedited Review Value for the relevant Disease Level;
 - (b) in Individual Review cases, the Maximum Value for the relevant Disease Level.
- 2.14.10 So far as the law permits, no Trust Claimant shall be entitled to challenge the validity of any act done or omitted to be done in good faith by the Expert in pursuance of his functions or duties, nor the exercise by the Expert in good faith of any power or discretion conferred upon him, and the Expert shall not be liable for any loss whatsoever and howsoever arising unless such loss is attributable to his own negligence, fraud or dishonesty.
- 2.14.11 The costs of the Expert shall be borne:
 - (a) Where the Trustees have rejected a Trust Claim and the Expert determines that the Trust Claim shall be admitted as an Established Claim, by the Trustees as an expense of the Trust.
 - (b) Where the Trustees have admitted a Trust Claim and the Expert determines that the Trust Claim shall be admitted as an Established Claim in an amount at least 20% higher than the amount determined by the Trustees, by the Trustees as an expense of the Trust.
 - (c) Where the Trustees have rejected a Trust Claim and the Expert determines that the Trust Claim shall be rejected, by the Trustees, who will be entitled to recover the amount of those costs from the Trust Claimant.

- (d) Where the Trustees have admitted a Trust Claim as an Established Claim and the Expert determines that the Trust Claim shall be admitted, but not for an amount at least 20% higher than the amount determined by the Trustees, by the Trustees, who will be entitled to deduct the costs from the amount due to the Trust Claimant up to the amount determined by the Expert, and any balance shall be an expense of the Trust.
- 2.14.12 The Trustees and the Trust Claimant shall each bear their own costs and disbursements of the referral of the matter to the Expert. In the case of the Trustees such costs and disbursements shall be expenses of the Trust.

SECTION THREE

3 Payments to Trust Claimants from the T&N Fund and the T&N Hercules Fund

The Trustees shall make payments to Trust Claimants from the T&N Fund and the T&N Hercules Fund in accordance with this Section 3. Clauses 3.1 to 3.6 deal with payments from the T&N Fund. Clause 3.7 deals with payments from the T&N Hercules Fund.

3.1 Computation of the T&N Payment Percentage; Reviews

- 3.1.1 The Trustees shall set the T&N Payment Percentage at such rate as they consider appropriate on the basis of the information and advice referred to in Clause 3.1.2, and after such reserves have been created by them for future claims as they consider to be prudent (for which purpose the Trustees shall be entitled in their absolute discretion to rely on actuarial projections based on a prudent best estimate), to enable them to comply with the provisions of this TDP in accordance with the Core Objective.
- 3.1.2 In determining the rate at which the T&N Payment Percentage is set, the Trustees shall consider and rely upon such legal, medical, fiscal, actuarial or other advice as they consider appropriate.
- 3.1.3 On the fifth anniversary of the Effective Date and every five years thereafter, and at such other times as they consider necessary, the Trustees shall review and if appropriate revise the T&N Payment Percentage as provided by Clauses 6.1 and 6.3.

3.2 Suspension of Payments from the T&N Fund

3.2.1 The Trustees shall suspend payments from the T&N Fund for such period (not exceeding three months) as they consider appropriate if information becomes available to them at any time as a result of which they determine that they should consider whether or not to set a revised T&N Payment Percentage outside the five year cycle provided by Clauses 6.1 and 6.3. As soon as practicable during, and in any event at the end of the suspension period, the

Trustees shall review the T&N Payment Percentage and either set a revised T&N Payment Percentage or conclude that the T&N Payment Percentage need not be revised, and thereupon the suspension of payments shall be lifted. A Payment Percentage set in accordance with this Clause shall be reviewed in accordance with the five year review cycle provided by Clauses 6.1 and 6.3.

3.3 Additional Dividend

3.3.1 Five years after making a payment to a Trust Claimant in respect of an Established Claim or as soon as reasonably practicable thereafter, the Trustees shall consider, in their absolute discretion, whether to pay the Trust Claimant an Additional Dividend. For the avoidance of doubt, if the Trustees determine that they shall pay an Additional Dividend on an Established Claim, there shall be only one such payment.

Value of Established Claims for the purpose of Additional Dividend

3.3.2 When considering the amount to be paid as an Additional Dividend, the Trustees may revise the value of any Established Claim they previously determined to take account of developments in the law and medical science in respect of the CVA Asbestos Claim the proceeds of which are assigned by the Trust Claimant to the Trust. The Trustees shall have an absolute discretion to revise the value of any Established Claim on these grounds.

Setting the level of Additional Dividend

- 3.3.3 The Trustees shall set the level of the Additional Dividend payable on Established Claims, first, within twelve months of the fifth anniversary of the Effective Date. Thereafter, the Trustees may review and if appropriate revise the level of the Additional Dividend every five years after the fifth anniversary of the Effective Date, as provided by Clauses 6.1 and 6.3, and at such other times as they consider necessary. The level of the Additional Dividend applied to the Established Claim shall be the level in force at the time the Additional Dividend is paid.
- 3.3.4 The Trustees shall not pay any Additional Dividend unless they consider, on the basis of the information and advice referred to in Clause 3.3.5, and by reference to the Additional Dividend, that after:
 - they have otherwise complied with the provisions of Clause 3.4 in relation to all Established Claims against the T&N Fund as at the date on which they consider the matter; and
 - (b) such reserves have been created by them for future claims as they consider to be prudent (for which purpose the Trustees shall be entitled in their absolute discretion to rely on actuarial projections based on a best estimate);

the T&N Fund will be sufficient to enable them to comply with the provisions of this TDP in accordance with the Core Objective.

3.3.5 In considering whether to pay an Additional Dividend the Trustees shall consider and rely upon such legal, medical, fiscal, actuarial or other advice as they consider appropriate.

Suspension

3.3.6 The Trustees shall be entitled to suspend payments of the Additional Dividend from the T&N Fund in accordance with Clause 3.2.1. For this purpose, references in Clause 3.2.1 to "payments" shall be replaced with "payments of the Additional Dividend", and references to "the T&N Payment Percentage" and "Payment Percentage" shall be replaced with "the level of the Additional Dividend".

3.4 Payments in respect of Established Claims from the T&N Fund

3.4.1 The Trustees shall calculate the amount to be paid by the Trustees out of the T&N Fund in respect of any Trust Claim which is admitted as an Established Claim in accordance with this Clause 3.4.

First Payment

3.4.2 The T&N Payment Percentage current at the date of payment shall be applied to the value of the Established Claim after giving credit for the actual payment of the T&N Hercules Payment Percentage out of the T&N Hercules Fund or the anticipated payment out of that Fund as estimated by the Trustees.

Additional Dividend

3.4.3 Where applicable, in accordance with Clause 3.3, the Additional Dividend shall be applied to the value of the Established Claim as determined by the Trustees.

Costs Payments

- 3.4.4 In addition to payments in respect of Established Claims from the T&N Fund or the T&N Hercules Fund (whichever makes payment first) the Trustees shall pay, on the occasion of the first such payment only, Costs Payments in respect of medical and legal costs in accordance with Clauses 3.4.5 and 3.4.6. Costs Payments shall be paid subject to Clause 2.4.4. As to medical and legal costs incurred by EL Scheme Trust Claimants, see Clause 3.4.9.
- 3.4.5 Subject to Clause 3.4.4, the Trustees shall pay from the T&N Fund to a Trust Claimant with an Established Claim against the T&N Fund as a Costs Payment such amount as the Trustees see fit to cover the provision of medical evidence up to a maximum of:

- (a) £750 £600 300 in cases of Disease Levels I and II;
- (b) £1,000 £800 400 in cases of Disease Levels II, III and IV;
- (c) $\frac{\text{\pounds}55}{\text{\pounds}50}$ in cases of Disease Level V.
- 3.4.6 Subject to Clause 3.4.4 above, the Trustees shall pay to a Trust Claimant with an Established Claim against the T&N Fund as a Costs Payment such amount as the Trustees see fit to cover the Trust Claimant's Actual Legal Costs up to a maximum of:
 - (a) $\frac{\pounds 650}{\pounds 575} \frac{\pounds 500}{\pounds 575}$ in relation to the first £5,000 of the Established Claim; plus
 - (b) 5% of the amount by which the Established Claim exceeds £5,000 but does not exceed £20,000; plus
 - (c) 2.5% of the amount by which the Established Claim exceeds £20,000; plus
 - (d) where the Trust Claimant has reasonably (in the view of the Trustees in their absolute discretion) issued proceedings in respect of the Trust Claim, the issue fee paid by the Trust Claimant in such proceedings;

subject to an overall maximum of $\frac{\pounds6,500}{\pounds5,750}$ $\frac{\pounds5,750}{\pounds5,000}$.

- 3.4.7 In order to determine the Actual Legal Costs the Trustees shall be entitled to require the Trust Claimant to provide a certificate signed by a partner or principal in the firm or firms of solicitors instructed by him setting out the costs and disbursements actually incurred on his behalf.
- 3.4.8 Any payment which is to be made in respect of Actual Legal Costs under Clauses 3.4.6 and 3.4.7 shall be paid by separate cheque directly in favour of the Trust Claimant's solicitors as nominated or identified by him in the Proof of Claim provided the name and other contact details including postal addresses of the solicitors so nominated are clearly stated.
- 3.4.9 EL Scheme Trust Claimants whose underlying EL Scheme Claims were admitted by the EL Scheme Trustees shall not be entitled to any payment for medical or legal costs incurred in respect of Established Claims against the T&N Fund.

TDP Determines Payments

3.4.10 Trust Claimants shall not be entitled to any payments from the Trust other than those provided for by this TDP.

3.5 Payment Mechanics

- 3.5.1 Subject to Clause 3.4.8, all sums payable to a Trust Claimant may be paid:
 - (a) by cheque in the name of the Trust Claimant or as he may direct in writing and may be sent by post to the Trust Claimant's last known address or such other address as is notified in writing by the Trust Claimant to the Trustees, or
 - (b) by telegraphic transfer to such bank account as the Trust Claimant may notify in writing to the Trustees.
- 3.5.1A Subject to clauses 3.5.2 and 3.5.3 below if, at the time of seeking to make a payment to a Trust Claimant, the Trustees are unsure as to whether they have the Trust Claimant's current address, the Trustees may in their absolute discretion employ agents on such terms as they consider appropriate to locate the Trust Claimant. The Trustees shall deduct the cost of employing any such agents from the payment to the Trust Claimant.
- 3.5.2 The sending of a cheque by the Trustees to the last known address of the Trust Claimant or his nominated solicitors, in accordance with Clause 3.5.1 (a) or Clause 3.4.8 as the case may be, shall be a good discharge of the Trustees and the Trustees shall not be liable to the Trust Claimant for any loss in transmission of a cheque drawn and sent in accordance with this Clause or Clause 3.4.8. Receipt of the amount of a telegraphic transfer into the Trust Claimant's bank account notified to the Trustees in accordance with Clause 3.5.1 (b) shall be a good discharge of the Trustees.
- 3.5.3 Where a telegraphic payment to a Trust Claimant or an unbanked cheque is returned to the Trustees unclaimed, or a cheque is unbanked for six months after its date, the Trustees shall transfer-retain in the payment account the amount of the unclaimed payment to a separate interest bearing account for unclaimed payments until 14 days before the date when the Trustees propose to terminate the Trust or one year after the payment was sent, whichever is the sooner, at which time, if the payment is still unclaimed, the Trustees shall transfer the unclaimed payment together with any interest earned to the Trust Claimant shall have no remaining right to them-or to any payment in respect of his Established Claim. If a Trust Claimant has lost the right to such unclaimed payment this shall not affect his right to receive any subsequent payments in respect of his Established Claim.

3.6 Deductions from Payments and Repayments

First payments in respect of Established Claims

3.6.1 The first payment to a Trust Claimant in respect of his Established Claim shall be made in full without any deduction, set-off, lien or withholding whatsoever save that:

- (a) The Trustees may deduct any costs incurred by the Expert from payments to Trust Claimants, as provided in Clause 2.14.11.
- (b) In the case of Established Claims assessed by Individual Review, the Trustees may set-off any amounts due in respect of a CRU Trust Claim which may lawfully be setoff against any amount due to the Trust Claimant. There shall be no such set-off in the case of Established Claims assessed by Expedited Review.
- (c) The Trustees may make such deduction on account of withholding or other tax as they may consider is required by law.

Repayment: no clawback except in cases of fraud

3.6.2 A Trust Claimant shall not be required to repay any sum that he received in respect of an Established Claim (whether a first payment or an Additional Dividend) unless the Trustees are satisfied that a fraud (including any fraudulent misrepresentation or non-disclosure) was committed by or on behalf of the Trust Claimant in respect of the Trust Claim. For the avoidance of doubt, a Trust Claimant shall not be required to repay any sum that he received in respect of an Established Claim (whether a first payment or an Additional Dividend) by virtue only of a reduction of the Payment Percentage applicable to his Established Claim.

Revising the value of the Established Claim

3.6.3 When considering the amount to be paid as an Additional Dividend, the Trustees may revise the value of any Established Claim they previously determined to take account of developments in the law and medical science which affect the CVA Asbestos Claim the proceeds of which are assigned by the Trust Claimant to the Trust. The Trustees shall have an absolute discretion to revise the value of any Established Claim on these grounds.

Overpayments and Additional Dividend

- 3.6.4 The Trustees shall be entitled to deduct any overpayment made to a Trust Claimant from any Additional Dividend due to such Trust Claimant out of the T&N Fund if:
 - (a) after a Trust Claim has been admitted and paid as an Established Claim, the Proof of Claim is withdrawn or varied as provided for in Clause 2.13.4; or
 - (b) after a Trust Claim has been admitted and paid as an Established Claim, it is shown to the satisfaction of the Trustees that an overpayment was made in respect of the Trust Claim by reason of any error, material misrepresentation or non-disclosure by or on behalf of the Trust Claimant.
- 3.6.5 The Trustees may also make such deduction from any Additional Dividend on account of withholding or other tax as they may consider is required by law.

3.7 The T&N Hercules Fund

3.7.1 The rights to payment from the Chester Street Fund and the Chester Street Hercules Fund pursuant to Clauses 2.4.3(b) and 4.1.2(c) of the Trust Deed and Clause 4.10 of this TDP shall constitute the T&N Hercules Fund, which shall be established on the date when the amendments to those Clauses implemented by the 2016(ii) (Tenth Anniversary) Amending Deed come into effect.

The UK Asbestos Trust Percentage of Hercules Recoveries received by the Trustees shall constitute the T&N Hercules Fund, which shall be established on the date when the Trustees receive the first payment in respect of Hercules Recoveries.

- 3.7.2 The Trustees shall pay a dividend from the T&N Hercules Fund to Trust Claimants with Established Claims against the T&N Fund <u>at the T&N Hercules Payment Percentage</u>, by reference to the quantum of Trust Claims that are made or are projected to be made.
- 3.7.2A Trust Claimants who have received a first dividend (but not any Additional Dividend) from the T&N Fund prior to the constitution of the T&N Hercules Fund shall be paid a first dividend from the T&N Hercules Fund at the T&N Hercules Payment Percentage once the T&N Hercules Fund has funds to distribute. Such Trust Claimants shall then be entitled to any Additional Dividend payable from the T&N Hercules Fund at the same time as any Additional Dividend is payable to them from the T&N Hercules Fund at the same time as any Additional Dividend is payable to them from the T&N Fund (or would be payable if there were sufficient funds available in the T&N Fund to pay an Additional Dividend). The Trustees may delay payment of a first dividend from the T&N Hercules Fund to a Trust Claimant until the payment of any Additional Dividend to that Trust Claimant if: (i) the Trustees consider such later payment to be in accordance with the Core Objective in view of the size of the first dividend; or (ii) the Trust Claimant does not initially respond to attempts by the Trustees to establish contact prior to the original payment date for the first dividend and contact is then established prior to the payment of any Additional Dividend.
- 3.7.2B Trust Claimants who have received a first dividend and an Additional Dividend from the T&N Fund prior to the constitution of the T&N Hercules Fund shall be paid a single dividend from the T&N Hercules Fund once the T&N Hercules Fund has funds to distribute. Such dividend shall be the sum of: (i) the T&N Hercules Payment Percentage applied to their Established Claim, and (ii) any Additional Dividend that the Trustees set pursuant to Clause 3.3 (as applied pursuant to this Clause 3.7) applied to their Established Claim.
- 3.7.2C Clauses 3.4.2 and 5.5.8(a) require the Trustees to apply the T&N Payment Percentage and any Non-T&N Fund Payment Percentage to the value of an Established Claim after giving credit for any actual payment from the T&N Hercules Fund, or after giving credit for any anticipated payment from the T&N Hercules Fund as estimated by the Trustees. Clauses 3.4.3 and 5.5.8(b) then require the Trustees to apply any Additional Dividend payable from the T&N Fund or Non-T&N Fund to the value of the Established Claim as determined by the

Trustees (and accordingly after giving credit for actual or anticipated payments from the T&N Hercules Fund). If:

- (a) the Trustees' Clause 3.4.2/3.4.3/5.5.8(a)/5.5.8(b) estimate of the anticipated payment out of the T&N Hercules Fund turns out to have been lower than the actual amount(s) that the Trustees later propose paying to a Trust Claimant from the T&N Hercules Fund on account of the T&N Hercules Payment Percentage and any Additional Dividend; and
- (b) <u>such actual proposed payment from the T&N Hercules Fund would result in a Trust</u> <u>Claimant recovering more in aggregate from the relevant Funds than they would</u> <u>otherwise have recovered if the Trustees' estimate had been correct (such additional</u> <u>amount being "the surplus"), then:</u>
 - (i) where the Trust Claimant has a Trust Claim against the T&N Fund and does not have an Agency (CVA Asbestos) Trust Claim, the Trustees shall pay the surplus from the T&N Hercules Fund to the T&N Fund; and
 - (ii) where the Trust Claimant has an Agency (CVA Asbestos) Trust Claim, the Trustees shall pay the surplus from the T&N Hercules Fund to the T&N Fund and the relevant Non-T&N Fund pro-rata to the original dividends paid out of those two Funds.
- 3.7.3 Clauses 3.1, 3.2, 3.3, <u>3.4.3</u>, <u>3.5</u>, and <u>3.6</u> shall apply to payments from the T&N Hercules Fund with the following modifications:
 - (a) "T&N Hercules Fund" shall be substituted for "T&N Fund";
 - (b) "T&N Hercules Payment Percentage" shall be substituted for "T&N Payment Percentage".
 - (c) As for Clause 3.1.1, for so long as the Trustees consider that the T&N Hercules Fund has sufficient value to permit it, the Trustees shall set the T&N Hercules Payment Percentage at such rate as they consider they would have set it if: (i) the US Asbestos Trust had not assigned its rights pursuant to Clauses 2.4.3(b) and 4.1.2(c) of the Trust Deed; (ii) the US Asbestos Trust had agreed pursuant to Paragraph 19.10 of the CVAs to change the UK Asbestos Trust Percentage and the Chester Street Percentage (between themselves) to such amounts as the Trustees considered necessary in order to promote a fair division of Hercules Recoveries as between the T&N Hercules Fund and the Chester Street Hercules Fund; (iii) the T&N Hercules Fund had been constituted by the UK Asbestos Trust Percentage (as varied) of the Hercules Recoveries payable pursuant Paragraph 19.8 of the CVAs following the commutation of the Hercules Policy on 28 October 2015; and (iv) there was to be no Additional Dividend payable from the T&N Hercules Fund.

- (d) As for Clause 3.1.3, the first review of the T&N Hercules Payment Percentage shall take place at the next Review Date in accordance with the five year cycle. Thereafter, the T&N Hercules Payment Percentage shall be reviewed every five years on the anniversary of the Effective Date, as provided by Clauses 6.1 and 6.3.
- (e) As for Clause 3.3.3, the Trustees shall first set the level of Additional Dividend payable on Established Claims from the T&N Hercules Fund within twelve months of the establishment of the T&N Hercules Fund. Thereafter, the Trustees may review and if appropriate revise the level of Additional Dividend in accordance with the five year cycle provided by Clauses 6.1 and 6.3.
- (f) As for Clause 3.3 generally, in determining the rate of any Additional Dividend the Trustees shall assume that the T&N Hercules Fund is constituted by the net present value of the whole of the share of the projected surplus in the Chester Street Fund and Chester Street Hercules Fund to which it is (or is projected to be) entitled pursuant to Clauses 2.4.3(b) and 4.1.2(c) of the Trust Deed and Clause 4.10 of this TDP.

SECTION FOUR

4 Chester Street Trust Claims

4.1 Assertion of Chester Street Trust Claims

Introduction

4.1.1 Chester Street Trust Claimants may assert their Trust Claims in the manner provided in Section 2, save for the exceptions in Clauses 4.1.2 to 4.1.6, which arise because this Section 4 makes special provision for Chester Street Trust Claims.

Exceptions for Chester Street Trust Claims

- 4.1.2 Chester Street Trust Claims may only proceed by way of Individual Review under Clause 2.6, and accordingly Clause 2.5 shall not apply.
- 4.1.3 Clauses 2.6.2, 2.6.3 (a), 2.6.6, 2.6.8 (b) and (d), 2.6.9, 2.6.10, 2.6.11, 2.6.14, 2.6.15 and 2.10.2 shall not apply to Chester Street Trust Claims.
- 4.1.4 If the Trustees determine that a Chester Street Trust Claim should be admitted as an Established Claim, they shall value the Established Claim as follows:

- (a) Where the Established Claim concerns a right to a contribution or indemnity, the Trustees may apply the Apportionment Formula, or such percentage as they consider appropriate or agree with the Chester Street Trust Claimant to:
 - the amount paid by the Chester Street Trust Claimant to the Injured Person, where this amount does not exceed the Maximum Value for the relevant Disease Level listed in Schedule 3; or
 - (ii) the Maximum Value for the relevant Disease Level listed in Schedule 3, where the amount paid by the Chester Street Trust Claimant to the Injured Person does exceed such Value.

(For the avoidance of doubt, Clauses 2.6.5, 2.6.6 and 2.6.7 shall not apply to the valuation of such Established Claims.)

- (b) Where the Established Claim is held by an Injured Person, the Trustees shall value it:
 - (i) in the manner provided by Clause 2.6.5 if the Injured Person holds a Pre-Administration Liquidated Claim; or
 - (ii) in all other cases in the manner provided by Clause 2.6.7.
- (c) Chester Street Trust Claimants shall not be entitled to any Costs Payments.
- 4.1.5 If the Trustees determine that a Chester Street Trust Claim should be admitted as an Established Claim, they shall send the Trust Claimant an Individual Review Admission Notice setting out in summary form the payment to be made to the Trust Claimant, showing the information required by Clause 2.6.8 (a) and (c). For this purpose references in Clause 2.6.8 (c) to the "T&N Payment Percentage" and the "T&N Hercules Payment Percentage" shall be replaced by references to the "Chester Street Payment Percentage" and the "Chester Street Hercules Payment Percentage".
- 4.1.6 If a Chester Street Trust Claimant refers the Trustees' decision as to the value of his Established Claim to the Expert in accordance with Clause 2.14, the Expert shall calculate the amount at which any Trust Claim is to be admitted as an Established Claim in accordance with Clause 4.1.4.

Specific application of Section 2

- 4.1.7 In the light of Clauses 4.1.1 to 4.1.6, the following provisions of Section 2 of this TDP shall apply with all necessary modifications to Chester Street Trust Claims:
 - (a) Clause 2.1 (advertising for Trust Claims).
 - (b) Clause 2.2 (time limits on Trust Claims).

- (c) Clause 2.3 (Proofs of Claim).
- (d) Clauses 2.6.1, 2.6.3 (b) to (f), 2.6.4, 2.6.5 (as provided by Clause 4.1.4 (b) (i)), 2.6.7 (as provided by Clause 4.1.4 (b) (ii)), 2.6.8 (a) and (c) (as provided by Clause 4.1.5), 2.6.12 and 2.6.13 (Individual Review Process).
- (e) Clause 2.7 (Trust Claims must be for the most severe Disease Level).
- (f) Clause 2.8 (Subsequent Trust Claims).
- (g) Clause 2.9 (special provisions applicable to deceased Injured Persons and certain living Injured Persons).
- (h) Clauses 2.10.1 and 2.10.3 (CRU Trust Claims), save that references in Clause 2.10.1 to the "T&N Fund", the "T&N Hercules Fund" and the "T&N Payment Percentage" shall be replaced by references to the "Chester Street Fund", the "Chester Street Hercules Fund", and the "Chester Street Payment Percentage".
- (i) Clause 2.11 (special provisions applicable to joint liability Trust Claims).
- (j) Clauses 2.12 (Trust Claims: co-operation of T&N and the Companies) and 2.13 (Trustees' powers and procedures).
- (k) Clause 2.14 (expert determination of disputes, subject to Clause 4.1.6).

4.2 Payments to Chester Street Trust Claimants from the Chester Street Fund and the Chester Street Hercules Fund

4.2.1 The Trustees shall make payments to Trust Claimants from the Chester Street Fund and the Chester Street Hercules Fund in accordance with this Section 4. Clauses 4.3 to 4.8 deal with payments from the Chester Street Fund. Clause 4.9 deals with payments from the Chester Street Fund. Clause 4.9 deals with payments from the Chester Street Hercules Fund. For the avoidance of doubt, Chester Street Trust Claimants with Established Claims shall not be entitled to any payments from any of the other Trust Funds.

4.3 Computation of the Chester Street Payment Percentage; Reviews

- 4.3.1 The Trustees shall pay Chester Street Trust Claimants with Established Claims at a percentage rate called the Chester Street Payment Percentage. This shall be set:
 - (a) For Chester Street Trust Claimants who held CVA Asbestos Claims against T&N, at a rate which gives a return equivalent to that which would have been received by T&N Trust Claimants if there had been no Costs Payments to T&N Trust Claimants under Clauses 3.4.4 to 3.4.7.
 - (b) For Chester Street Trust Claimants who held CVA Asbestos Claims against any Company other than T&N (except Agency Companies), at a rate which gives a return

equivalent to that which would have been received by Non-T&N Trust Claimants if there had been no Costs Payments to Non-T&N Trust Claimants under Clauses 5.2.1 to 5.2.4. This rate shall be calculated by reference to either:

- the dividend paid on Non-T&N Fund Claims by the Supervisor of the relevant Company's CVA; or
- (ii) after a transfer has been made pursuant to paragraphs 5.7.1 to 5.7.3 of the CVAs, by reference to the Payment Percentage of the Non-T&N Fund concerned.
- (c) For Chester Street Trust Claimants who held CVA Asbestos Claims against Agency Companies (except Newalls Insulation Company Limited, TBA Belting Limited, T&N Shelf Twenty-Six Limited, The Washington Chemical Company Limited, or JW Roberts Limited), at a rate which gives a return equivalent to the aggregate of:
 - the rate which would have been received by T&N Trust Claimants if there had been no Costs Payments to T&N Trust Claimants under Clauses 3.4.4 to 3.4.7; and
 - (ii) the rate which would have been received by Non-T&N Trust Claimants if there had been no Costs Payments to Non-T&N Trust Claimants under Clauses 5.2.1 to 5.2.4. This rate shall be calculated by reference to either:
 - (aa) the dividend paid on Non-T&N Fund Claims by the Supervisor of the relevant Company's CVA; or
 - (bb) after a transfer has been made pursuant to paragraphs 5.7.1 to 5.7.3 of the CVAs, by reference to the Payment Percentage of the Non-T&N Fund concerned,

provided that such Chester Street Trust Claimants will not be entitled in any circumstances to recover more than 100% of the value of their Established Claims.

- (d) For Chester Street Trust Claimants who held CVA Asbestos Claims against Newalls Insulation Company Limited, TBA Belting Limited, T&N Shelf Twenty-Six Limited, The Washington Chemical Company Limited, or JW Roberts Limited, at a rate which gives a return equivalent to the aggregate of:
 - the rate which would have been received by T&N Trust Claimants if there had been no Costs Payments to T&N Trust Claimants under Clauses 3.4.4 to 3.4.7; and
 - (ii) a fixed sum payment (as the case may be) of:

- (aa) £5 for Newalls Insulation Company Limited and TBA Belting Limited;
- (bb) £60 for T&N Shelf Twenty-Six Limited;
- (cc) £30 for The Washington Chemical Company Limited.;

(dd) £20 for JW Roberts Limited.

- 4.3.2 In circumstances where the Trustees revise the T&N Payment Percentage in accordance with Clause 3.1.3, they shall revise the Chester Street Payment Percentage to the extent necessary to achieve the equivalence required by Clauses 4.3.1 (a), 4.3.1 (c) (i), and 4.3.1 (d) (i).
- 4.3.3 In circumstances where the Trustees revise the Non-T&N Payment Percentage which they set after a transfer has been made pursuant to Clauses 5.7.1 to 5.7.3 of the CVAs, they shall revise the Chester Street Payment Percentage to the extent necessary to achieve the equivalence required by Clauses 4.3.1 (b) (ii) and 4.3.1 (c) (ii) (bb).
- 4.4 Suspension of Payments from the Chester Street Fund and compromise of Trust Claims
- 4.4.1 If the Trustees suspend payments from the T&N Fund in the circumstances set out in Clause 3.2.1, they shall suspend payments from the Chester Street Fund where the Chester Street Payment Percentage is calculated in accordance with Clauses 4.3.1 (a), 4.3.1 (c) (i) or 4.3.1 (d) (i). When the Trustees lift the suspension of payments from the T&N Fund (having either set a revised T&N Payment Percentage or concluded that the T&N Payment Percentage need not be revised in accordance with Clause 3.2.1), they shall revise the Chester Street Payment Percentage if necessary and they shall lift any suspension of payments from the Chester Street Fund. A Chester Street Payment Percentage revised in accordance with the five year review cycle as provided by Clauses 6.1 and 6.4.
- 4.4.2 If the Trustees suspend payments from a Non-T&N Fund after a transfer has been made pursuant to paragraphs 5.7.1 to 5.7.3 of the CVAs, they shall suspend payments from the Chester Street Fund where the Chester Street Payment Percentage is calculated in accordance with Clauses 4.3.1 (b) (ii) or 4.3.1 (c) (ii) (bb). When the Trustees lift the suspension of payments from the Non-T&N Fund concerned (having either set a revised Payment Percentage or concluded that the Payment Percentage need not be revised in accordance with Clause 5.5.6), they shall revise the Chester Street Payment Percentage if necessary and they shall lift any suspension of payments from the Chester Street Fund. A Chester Street Payment Percentage revised in accordance with the five year review cycle as provided by Clauses 6.1 and 6.4.

4.4.3 For the avoidance of doubt, the Trustees may compromise Chester Street Trust Claims if they consider it appropriate by agreement with the relevant Chester Street Trust Claimant.

4.5 Additional Dividend

- 4.5.1 If the Trustees pay an Additional Dividend in respect of Established Claims against the T&N Fund in accordance with Clause 3.3, they shall pay a corresponding Additional Dividend in relation to Established Claims against the Chester Street Fund, where the Chester Street Payment Percentage is calculated in accordance with Clauses 4.3.1 (a), 4.3.1 (c) (i), or 4.3.1 (d) (i).
- 4.5.2 If the Trustees pay an Additional Dividend in respect of Established Claims against any of the Non-T&N Funds after a transfer has been made under paragraphs 5.7.1 to 5.7.3 of the CVAs, they shall pay a corresponding Additional Dividend in relation to Established Claims against the Chester Street Fund, where the Chester Street Payment Percentage is calculated in accordance with Clauses 4.3.1 (b) (ii) or 4.3.1 (c) (ii) (bb).
- 4.5.3 When considering the amount to be paid as an Additional Dividend under Clauses 4.5.1 or 4.5.2, the Trustees may revise the value of any Established Claim they previously determined to take account of developments in the law or medical science which affect the CVA Asbestos Claim the proceeds of which are assigned by the Trust Claimant to the Trust. The Trustees shall have an absolute discretion to revise the value of any Established Claim on these grounds.
- 4.5.4 Where the Trustees pay an Additional Dividend, every five years from the fifth anniversary of the Effective Date and at such other times as they consider necessary, they shall review and if appropriate revise the level of Additional Dividend, as provided by Clauses 6.1 and 6.4 below.

4.6 Payments in respect of Established Claims

4.6.1 The Trustees shall calculate the amount to be paid by the Trustees out of the Chester Street Fund in respect of any Chester Street Trust Claim which is admitted as an Established Claim in accordance with this Clause 4.6.

First Payment

4.6.2 The Chester Street Payment Percentage current at the date of payment shall be applied to the value of the Established Claim as determined by the Trustees after giving credit for the actual payment of the Chester Street Hercules Payment Percentage out of the Chester Street Hercules Fund, or the anticipated payment out of that Fund as estimated by the Trustees.

Additional Dividend

4.6.3 Where applicable in accordance with Clause 4.5, the Additional Dividend shall be applied to the value of the Established Claim as determined by the Trustees.

Payment(s) in accordance with Clause 4.6 and from the Chester Street Funds

4.6.4 Chester Street Trust Claimants shall not receive any payments from the Trust other than those provided for by this Clause 4.6, and shall not in any event be entitled to receive any payment from any of the other Trust Funds.

4.7 Payment Mechanics

4.7.1 Clause 3.5 shall apply to payments to be made to Chester Street Trust Claimants with Established Claims.

4.8 Deductions from Payments and Repayments

4.8.1 Clause 3.6 applies to payments to Chester Street Claimants, save that there shall be no setoff in respect of any amounts due in relation to a CRU Trust Claim as provided by Clause 3.6.1 (b).

4.9 The Chester Street Hercules Fund

4.9.1 The rights to payment from the Chester Street Fund pursuant to Clauses 2.4.3(b) and 4.1.2(c) of the Trust Deed and Clause 4.10 of this TDP shall constitute the Chester Street Hercules Fund, which shall be established on the date when the amendments to those Clauses implemented by the 2016(ii) (Tenth Anniversary) Amending Deed come into effect.

The Chester Street Percentage of Hercules Recoveries received by the Trustees shall constitute the Chester Street Hercules Fund, which shall be established on the date when the Trustees receive the first payment in respect of Hercules Recoveries.

- 4.9.2 The Trustees shall pay a dividend from the Chester Street Hercules Fund to Chester Street Trust Claimants with Established Claims, save that Chester Street Trust Claimants who held CVA Asbestos Claims against Companies other than T&N which are not Agency Companies will not be entitled to any such dividend. The dividend shall be equivalent to the dividend paid from the T&N Hercules Fund to Trust Claimants with Established Claims against the T&N Fund and will be "the Chester Street Hercules Payment Percentage".
- 4.9.2A Trust Claimants who have received a first dividend (but not any Additional Dividend) from the Chester Street Fund prior to the constitution of the Chester Street Hercules Fund shall be paid a first dividend from the Chester Street Hercules Fund at the Chester Street Hercules Payment Percentage once the Chester Street Hercules Fund has funds to distribute. Such Trust Claimants shall then be entitled to any Additional Dividend payable from the Chester

Street Hercules Fund at the same time as any Additional Dividend is payable to them from the Chester Street Fund (or would be payable if there were sufficient funds available in the Chester Street Fund to pay an Additional Dividend). The Trustees may delay payment of a first dividend from the Chester Street Hercules Fund to a Trust Claimant until the payment of any Additional Dividend to that Trust Claimant if: (i) the Trustees consider such later payment to be in accordance with the Core Objective in view of the size of the first dividend; or (ii) the Trust Claimant does not initially respond to attempts by the Trustees to establish contact prior to the original payment date for the first dividend and contact is then established prior to the payment of any Additional Dividend.

- 4.9.2B Trust Claimants who have received a first dividend and an Additional Dividend from the Chester Street Fund prior to the constitution of the Chester Street Hercules Fund shall be paid a single dividend from the Chester Street Hercules Fund once the Chester Street Hercules Fund has funds to distribute. Such dividend shall be the sum of: (i) the Chester Street Hercules Payment Percentage applied to their Established Claim, and (ii) any Additional Dividend that the Trustees set pursuant to Clause 4.5 (as applied pursuant to this Clause 4.9) applied to their Established Claim.
- 4.9.2C Clause 4.6.2 requires the Trustees to apply the Chester Street Payment Percentage to the value of an Established Claim after giving credit for any actual payment from the Chester Street Hercules Fund, or after giving credit for any anticipated payment from the Chester Street Hercules Fund as estimated by the Trustees. Clause 4.6.3 then requires the Trustees to apply any Additional Dividend payable from the Chester Street Fund to the value of the Established Claim as determined by the Trustees (and accordingly after giving credit for actual or anticipated payments from the Chester Street Hercules Fund). If:
 - (a) the Trustees' Clause 4.6.2/4.6.3 estimate of the anticipated payment out of the Chester Street Hercules Fund turns out to have been lower than the actual amount(s) that the Trustees later propose paying to a Trust Claimant from the Chester Street Hercules Fund on account of the Chester Street Hercules Payment Percentage and any Additional Dividend; and
 - (b) <u>such actual proposed payment from the Chester Street Hercules Fund would result in</u> <u>a Trust Claimant recovering more in aggregate from the relevant Funds than they</u> <u>would otherwise have recovered if the Trustees' estimate had been correct (such</u> <u>additional amount being "the surplus"),</u>

then the Trustees shall pay the surplus from the Chester Street Hercules Fund to the Chester Street Fund.

4.9.3 Clauses 4.4, 4.5, <u>4.6.3</u>, 4.7, and 4.8 shall apply to payments from the Chester Street Hercules Fund with the following modifications:

- (a) "Chester Street Hercules Fund" shall be substituted for "Chester Street Fund".
- (b) "T&N Hercules Fund" shall be substituted for "T&N Fund".
- (c) "Chester Street Hercules Payment Percentage" shall be substituted for "Chester Street Payment Percentage".
- (d) "T&N Hercules Payment Percentage" shall be substituted for "T&N Payment Percentage".
- (e) In Clause 4.4.1 the Chester Street Hercules Payment Percentage will have been calculated in accordance with Clause 4.9.2 only.
- (f) Clause 4.4.2 shall not apply.
- (g) In Clause 4.5.1 the Chester Street Hercules Payment Percentage will have been calculated in accordance with Clause 4.9.2 only.
- (h) Clause 4.5.2 shall not apply.

4.10 Payments to the T&N Hercules Fund and Chester Street Hercules Fund pursuant to Clauses 2.4.3(b) and 4.1.2(c) of the Trust Deed

- 4.10.1 Upon the constitution of the T&N Hercules Fund and the Chester Street Hercules Fund, the Trustees shall determine whether there is a surplus of Chester Street Fund Assets in either the Chester Street Fund or the Chester Street Hercules Fund beyond those that they project will be required for the payment of the holders of Chester Street Trust Claims pursuant to Clauses 2.4.3(a) or 4.1.2(a) or (b) of the Trust Deed or for the payment of any costs payable from the Chester Street Fund or Chester Street Hercules Fund pursuant to Clauses 17 or 18 of the Trust Deed (or otherwise), as applicable,
- 4.10.2 For the purposes of determining whether there is a surplus pursuant to Clause 4.10.1 above, the Trustees may in their absolute discretion rely on actuarial projections based on a prudent best estimate and upon such legal, medical, fiscal, actuarial or other advice as they consider appropriate.
- 4.10.3 <u>The Trustees shall review and if appropriate revise their determination of the surplus on every</u> <u>fifth anniversary of the Effective Date and at such other times as they consider necessary.</u>
- 4.10.4 For so long as the Trustees project a surplus in accordance with Clause 4.1.2(c) of the Trust Deed and this Clause 4.10, the Trustees shall on a monthly basis (and subject at all times to a cap of the net present value of the projected surplus) pay to the T&N Hercules Fund and the Chester Street Hercules Fund, according to their rights under Clause 2.4.3(b) of the Trust Deed:

- (a) such amounts as are required that month by the T&N Hercules Fund and the Chester Street Hercules Fund to pay the T&N Hercules Payment Percentage or Chester Street Hercules Payment Percentage (as applicable) on the value of Established Claims;
- (b) such amounts as are required that month by the T&N Hercules Fund and the Chester Street Hercules Fund to pay any applicable Additional Dividends on the value of Established Claims; and
- (c) <u>such amounts as are required that month by the T&N Hercules Fund and the Chester</u> <u>Street Hercules Fund to pay any costs that are payable pursuant to the Trust Deed or</u> <u>this TDP.</u>

SECTION FIVE

5 Payments in relation to Established Claims against the Non-T&N Funds

The Trustees shall follow the procedure set out below for payments in relation to Established Claims against the Non-T&N Funds.

5.1 Non-T&N Fund Claims

5.1.1 Having determined the value of an Established Claim in respect of a Company other than T&N, the Trustees shall submit a Non-T&N Fund Claim in the CVA of the relevant Company in accordance with Clauses 2.5.9 (b), 2.5.10 (b), 2.5.11 (b), 2.6.9 (b), 2.6.10 (b) and 2.6.11 (b), as the case may be.

5.2 Medical and legal costs

- 5.2.1 As part of a Non-T&N Fund Claim for the value of the Established Claim determined by the Trustees, the Trustees shall be entitled to claim in the CVA of the relevant Company medical and legal costs to the extent provided by Clauses 2.5.9 (b), 2.5.10 (b), 2.6.9 (b) and 2.6.10 (b), subject to Clause 2.4.4, and calculated in accordance with Clauses 5.2.2 and 5.2.3. As stated in Clauses 2.5.11 (b) and 2.6.11 (b), no Costs Payments shall be claimed as part of Non-T&N Fund Claims in the CVAs of Newalls Insulation Company Limited, TBA Belting Limited, T&N Shelf Twenty-Six Limited, The Washington Chemical Company Limited, or JW Roberts Limited. As to medical and legal costs incurred in respect of Established Claims concerning Companies other than T&N held by EL Scheme Trust Claimants, Clause 5.2.6 shall apply.
- 5.2.2 Subject to Clause 5.2.1 above, the Trustees shall claim in the CVA of the relevant Company such amount as the Trustees in their discretion see fit to cover the provision of medical evidence up to a maximum of:

- (a) £750 £600 300 in cases of Disease Levels I-and II;
- (b) £1,000 £800 400 in cases of Disease Levels II, III and IV;
- (c) $\frac{\text{\pounds}55}{\text{\pounds}50}$ in cases of Disease Level V.
- 5.2.3 Subject to Clause 5.2.1 above, the Trustees shall claim in the CVA of the relevant Company such amount as the Trustees in their discretion see fit to cover the Trust Claimant's Actual Legal Costs up to a maximum of:
 - (a) $\frac{\pounds 650}{\pounds 575} \frac{\pounds 500}{\pounds 575}$ in relation to the first £5,000 of the Established Claim; plus
 - (b) 5% of the amount by which the Established Claim exceeds £5,000 but does not exceed £20,000; plus
 - (c) 2.5% of the amount by which the Established Claim exceeds £20,000; plus
 - (d) where the Trust Claimant has reasonably (in the view of the Trustees in their absolute discretion) issued proceedings in respect of that Trust Claim, the issue fee paid by the Trust Claimant in such proceedings;

subject to an overall maximum of $\frac{\pounds6,500}{\pounds5,750}$ $\frac{\pounds5,750}{\pounds5,000}$.

- 5.2.4 In order to determine the Actual Legal Costs the Trustees shall be entitled to require the Trust Claimant to provide a certificate signed by a partner or principal in the firm or firms of solicitors instructed by him setting out the costs and disbursements actually incurred on his behalf.
- 5.2.5 Notwithstanding the provisions of Clause 5.3.1, that part of any Costs Payment which is to be made in respect of Actual Legal Costs under Clauses 5.2.3 and 5.2.4, shall be paid by separate cheque directly in favour of the Trust Claimant's solicitors as nominated or identified by him in the Proof of Claim provided that the name and other contact details including postal addresses of the solicitors so nominated are clearly stated.
- 5.2.6 EL Scheme Trust Claimants whose underlying EL Scheme Claims were admitted by the EL Scheme Trustees shall not be entitled to any payments for medical or legal costs incurred in respect of Established Claims concerning Companies other than T&N.

5.3 Payment Mechanics

5.3.1 For the avoidance of doubt, Clause 3.5 shall apply in the case of Established Claims concerning Companies other than T&N.

5.4 **Deductions from Payments and Repayments**

5.4.1 Clauses 3.6.1 to 3.6.3 shall apply in the case of Established Claims concerning Companies other than T&N.

- 5.4.2 Without prejudice to any of their other rights to inform the relevant CVA Supervisor of any overpayment made to a Trust Claimant holding an Established Claim in respect of a Company other than T&N following the submission of a Non-T&N Fund Claim by the Trustees under Clauses 2.5.9 (b), 2.5.10 (b), 2.5.11 (b), 2.6.9 (b), 2.6.10 (b), or 2.6.11 (b), the Trustees shall be entitled to deduct any overpayment from any further sum or sums due from time to time to such Trust Claimant (including any Additional Dividend, in circumstances where a transfer has been made to the relevant Non-T&N Fund under paragraphs 5.7.1 to 5.7.3 of the CVAs) if:
 - (a) after a Trust Claim has been admitted as an Established Claim, and a Non-T&N Fund Claim has been submitted in the CVA of the relevant Company, the Proof of Claim is withdrawn or varied as provided for in Clause 2.13.4; or
 - (b) after a Trust Claim has been admitted as an Established Claim, and a Non-T&N Fund Claim has been submitted in the CVA of the relevant Company, it is shown to the satisfaction of the Trustees that the Trust Claim was admitted or the Non-T&N Claim was submitted in too high a sum by reason of any error, material misrepresentation or non-disclosure by or on behalf of the Trust Claimant.
- 5.4.3 The Trustees may also make such deduction from any further sum or sums due from time to time to such Trust Claimant (including any Additional Dividend, in circumstances where a transfer has been made to the relevant Non-T&N Fund under paragraphs 5.7.1 to 5.7.3 of the CVAs) on account of withholding or other tax as they may consider is required by law.

5.5 The operation of Non-T&N Funds after the completion of the relevant CVA

Transfers under paragraphs 5.7.1 to 5.7.3 of the CVAs

- 5.5.1 In the event that the Supervisors of any of the Companies pay any monies to the Trustees pursuant to paragraph 5.7.1 to 5.7.3 of the CVAs, the monies so paid shall form part of the Non-T&N Fund Assets in relation to the relevant Company, and shall be used to make such payments as are due under the Trust Deed and this TDP to holders of Non-T&N Trust Claims in respect of the Company.
- 5.5.2 Furthermore, in the event of any such payment:
 - (a) the monies shall be accepted by the Trustees in full and final settlement of all further Non-T&N Fund Claims which would or may otherwise have been made against the relevant Non-T&N Subreserve in the Company's CVA, such that no further Non-T&N Fund Claims shall be made in the CVA whether pursuant to paragraph 14.3 of the CVAs or otherwise;
 - (b) in the event that the sum paid to the Trustees (when aggregated with all other funds standing to the credit of the relevant Non-T&N Fund) is in excess of the value of all Non-T&N Trust Claims in respect of the relevant Company, the Trustees, upon

concluding that it is unlikely that any further Non-T&N Trust Claims against the relevant Company will be made:

- shall apply any surplus funds held in the Non-T&N Fund in payment of Non-T&N Trust Claims against other Companies; and
- (ii) to the extent not applied in accordance with the preceding sub-paragraph, shall pay any balance to the T&N Fund, provided that up to 22/55 of any sum received by the T&N Fund may be paid by the Trustees to the Chester Street Fund in the event (and only in the event) that the Chester Street Fund Assets (including any payments under this Clause and paragraphs 5.7.3) (d) and 5.8 of the CVAs) become exhausted and to the extent such payments are required to fund Chester Street Fund Costs and/or Established Claims against the Chester Street Fund.

Setting Payment Percentages for the Non-T&N Funds

- 5.5.3 Upon the payment to the Trust of any such monies, the Trustees shall set a Payment Percentage for the Non-T&N Fund concerned at such rate as they consider appropriate on the basis of the information and advice referred to in Clause 5.5.4, and after such reserves for future claims have been created by them out of the Non-T&N Fund concerned as they consider to be prudent (for which purpose the Trustees shall be entitled in their absolute discretion to rely on actuarial projections based on a prudent best estimate) to enable them to comply with the provisions of this TDP in accordance with the Core Objective.
- 5.5.4 In determining the rate at which such a Payment Percentage is set, the Trustees shall consider and rely upon such legal, medical, fiscal, actuarial or other advice as they consider appropriate.
- 5.5.5 The Trustees shall review and if appropriate revise the Payment Percentage for a Non-T&N Fund in respect of any of the Companies at the next review of the T&N Payment Percentage pursuant to Clauses 6.1 and 6.3, and every five years thereafter as provided by Clause 6.5, and at such other times as they consider necessary.

Suspension of Payments from the Non-T&N Funds

5.5.6 The Trustees may suspend payments from a Non-T&N Fund in respect of any of the Companies for such period (not exceeding three months) as they consider appropriate if information becomes available to them at any time as a result of which they determine that they should consider whether or not to set a revised Payment Percentage outside the five year cycle provided by Clause 6.5. As soon as practicable during, and in any event at the end of any suspension period, the Trustees shall review the Payment Percentage and either set a revised Payment Percentage or conclude that the Payment Percentage need not be revised and thereupon the suspension of payments shall be lifted. A Payment Percentage set in

accordance with this Clause shall be reviewed in accordance with the five year review cycle provided by Clauses 6.1 and 6.5.

Additional Dividend

- 5.5.7 The Trustees shall be entitled to pay an Additional Dividend from a Non-T&N Fund as follows:
 - (a) The process described in Clauses 3.3.1 to 3.3.5 shall apply save that:
 - (i) As for Clause 3.3.3, the Trustees shall first set the level of Additional Dividend payable on Established Claims within twelve months of the fifth anniversary of the Effective Date following the transfer of monies to the Trustees under paragraphs 5.7.1 to 5.7.3 of the CVAs; and
 - (ii) the reference in Clause 3.3.4 to "the T&N Fund" shall be replaced by a reference to the Non-T&N Fund concerned.
 - (b) The Trustees shall be entitled to suspend payments of the Additional Dividend in accordance with Clause 5.5.6. For this purpose, references in Clause 5.5.6 to "payments" shall be replaced with "payments of the Additional Dividend", and references to "the Payment Percentage" shall be replaced with "the level of the Additional Dividend".

Payments in respect of Established Claims from the Non-T&N Funds

- 5.5.8 Where the Trustees hold monies in a Non-T&N Fund after a transfer pursuant to paragraphs5.7.1 to 5.7.3 of the CVAs, they shall make payments for Established Claims in relation to the Company concerned as follows:
 - (a) Where appropriate, the Payment Percentage current at the date of payment shall be applied to the value of the Established Claim after giving credit for any actual payment of the T&N Hercules Payment Percentage out of the T&N Hercules Fund or any anticipated payment out of that Fund as estimated by the Trustees.
 - (b) Where applicable in accordance with Clause 5.5.7, the Additional Dividend shall be applied to the value of the Established Claim as determined by the Trustees.
 - (c) Payments may be made in respect of medical and legal costs in accordance with Clauses 3.4.4 to 3.4.7, and references therein to "the T&N Fund" shall be replaced by references to the Non-T&N Fund concerned.
 - (d) The following further provisions of Section 3 shall apply, with references to "the T&N Fund" replaced by references to the Non-T&N Fund concerned:

- (i) Clause 3.5 (Payment Mechanics).
- (ii) Clause 3.6 (Deductions from Payments and Repayments).

SECTION SIX

6 Revision of Values, maximum figures for pain and suffering, Payment Percentages, costs cap, levels of Additional Dividend and medical costs.

6.1 General Provisions

- 6.1.1 In order to comply with the requirements of the TDP and the Core Objective, on the fifth anniversary of the Effective Date and every five years thereafter, and at such other times as they consider necessary, the Trustees shall review and if appropriate revise either upwards or downwards, or keep at the same level:
 - (a) The Expedited Review Values and the Maximum Values in Schedule 3.
 - (b) The maximum figures representing the pain and suffering element of a Trust Claim payable in accordance with Clause 2.9.3.
 - (c) The T&N Payment Percentage in Clause 3.1 and the T&N Hercules Payment Percentage in Clause 3.7, together with the level of any Additional Dividend.
 - (d) The Chester Street Payment Percentage where it is calculated in accordance with Clauses 4.3.1 (a), 4.3.1 (b) (ii), 4.3.1 (c) (i), 4.3.1 (c) (ii) (bb) and 4.3.1 (d) (i), and the Chester Street Hercules Payment Percentage in Clause 4.9.2, together with the level of any Additional Dividend.
 - (e) Any Payment Percentage for a Non-T&N Fund established for any of the Companies other than T&N under Clause 5.5, together with the level of any Additional Dividend.
 - (f) The cap of £5,000 imposed on Actual Legal Costs in Clauses 3.4.6 and 5.2.3.
- 6.1.2 At any time the Trustees may in their absolute discretion review and if appropriate revise either upwards or downwards, or keep at the same level the payments for medical costs in Clauses 3.4.5 and 5.2.2.
- 6.1.3 In conducting their review, the Trustees shall in their absolute discretion consider all such matters as appear to them to be relevant to their review, and may rely upon such legal, medical, fiscal, actuarial or other advice as they consider appropriate.

- 6.1.4 The Trustees may in their absolute discretion postpone their 5-yearly review for no longer than 12 months.
- 6.1.5 If the Trustees postpone a review, the Expedited Review Values, Maximum Values, maximum figures for pain and suffering, Payment Percentages, cap on Actual Legal Costs, levels of Additional Dividend and medical costs (as the case may be) from the previous 5 year period will continue to apply until the review takes place. Any increases resulting from the review can be backdated for Trust Claims submitted during the period that the review was postponed, but decreases shall not be relied upon to claw back any sum already paid to a Trust Claimant.

6.2 Special provision for Values in Schedule 3

- 6.2.1 Where a Trust Claim is submitted to the Trust and allowed by the Trustees as an Established Claim either (i) before the first five yearly revision of the Expedited Review Values and Maximum Values, or (ii) after a revision has taken place, but before the next revision is carried out, Established Claims shall be valued as follows:
 - (a) Where the Established Claim is assessed by Expedited Review, the Trustees shall value it by the application of the Expedited Review Value for that five year period, subject to any discounts made pursuant to Clauses <u>2.5.6 and</u> 2.5.7.
 - (b) Where the Established Claim is assessed by Individual Review:
 - (i) As to any recovery for pain and suffering, the Trustees shall calculate the value of any such recovery by reference to the amount that could have been recovered as at the Effective Date or the most recent review date, as the case may be.
 - (ii) As to any recovery for other heads of loss, the Trustees may in their absolute discretion discount any such recovery by an amount for inflation calculated by reference to the cumulative rates contained in the Retail Price Index from the date on which the Maximum Values were last set to the date on which the Trust Claim was submitted to the Trust.

6.3 Special Provisions for the T&N Payment Percentage and the T&N Hercules Payment Percentage

- 6.3.1 The Trustees shall only set an increased T&N Payment Percentage or an increased T&N Hercules Payment Percentage pursuant to clause 6.1 if they consider, on the basis of the information and advice referred to in Clause 6.3.2, that after:
 - (a) they have complied with the provisions of Clause 3.3 in relation to all Established Claims against the T&N Fund as at the Review Date concerned; and

(b) such reserves have been created by them for future claims as they consider to be prudent (for which purpose the Trustees shall be entitled in their absolute discretion to rely on actuarial projections based on a best estimate);

the T&N Fund and the T&N Hercules Fund will be sufficient to enable them to comply with the provisions of this TDP in accordance with the Core Objective.

- 6.3.2 In reviewing the rate at which the T&N Payment Percentage and the T&N Hercules Payment Percentage are set, the Trustees shall consider and rely upon such legal, medical, fiscal, actuarial or other advice as they consider appropriate.
- 6.3.3 Clauses 6.3.1 and 6.3.2 apply to the level of any Additional Dividend that the Trustees determine should be paid in accordance with Clause 3.3.
- 6.3.4 This Clause 6.3 shall be without prejudice to the Trustees' powers under Clause 3.2.1 to suspend payments from the T&N Fund and review and if appropriate revise the T&N Payment Percentage, and Clause 3.7.3 which provides the same powers in relation to the T&N Hercules Fund and the T&N Hercules Payment Percentage.

6.4 Special Provisions for the Chester Street Payment Percentage and the Chester Street Hercules Payment Percentage

- 6.4.1 Where the Trustees revise the T&N Payment Percentage and the T&N Hercules Payment Percentage pursuant to Clauses 6.1 and 6.3, they shall revise the Chester Street Payment Percentage and the Chester Street Hercules Payment Percentage in so far as they are calculated in accordance with Clauses 4.3.1 (a), 4.3.1 (c) (i) or 4.3.1 (d) (i), and to the extent necessary to set a rate which gives a return equivalent to that which would have been received by T&N Trust Claimants if there had been no Costs Payments to T&N Trust Claimants under Clauses 3.4.4 to 3.4.7.
- 6.4.2 Where the Trustees revise the Payment Percentage for a Non-T&N Fund pursuant to Clauses 5.5.5 and 6.5, they shall revise the Chester Street Payment Percentage (but not the Chester Street Hercules Payment Percentage) in so far as it was calculated in accordance with Clauses 4.3.1 (b) (ii) or 4.3.1 (c) (ii) (bb), and to the extent necessary to set a rate which gives a return equivalent to that which would have been received by Non-T&N Trust Claimants if there had been no Costs Payments to Non-T&N Trust Claimants under Clauses 5.2.1 to 5.2.4.
- 6.4.3 Clauses 6.4.1 and 6.4.2 apply to the level of any Additional Dividend that the Trustees determine should be paid in accordance with Clause 4.5 above.
- 6.4.4 This Clause 6.4 shall be without prejudice to the Trustees' powers under Clauses 4.4.1 and 4.4.2 to suspend payments from the Chester Street Fund and to review and if appropriate revise the Chester Street Payment Percentage as there set out, and Clause 4.9.3, which

provides the same powers in relation to the Chester Street Hercules Fund and the Chester Street Hercules Payment Percentage.

6.5 Special Provisions for the Payment Percentages for Non-T&N Funds after the completion of the relevant CVA

- 6.5.1 The Trustees shall only set an increased Payment Percentage pursuant to Clause 6.1 if they consider, on the basis of the information and advice referred to in Clause 6.5.2, that after such reserves have been created by them for future claims as they consider to be prudent (for which purpose the Trustees shall be entitled in their absolute discretion to rely on actuarial projections based on a best estimate), the Fund will be sufficient to enable them to comply with the provisions of this TDP in accordance with the Core Objective.
- 6.5.2 In reviewing the rate at which any such Payment Percentage is set, the Trustees shall consider and rely upon such legal, medical, fiscal, actuarial or other advice as they consider appropriate.
- 6.5.3 Clauses 6.5.1 and 6.5.2 shall apply to the level of any Additional Dividend that the Trustees determine should be paid in accordance with Clause 5.5.7.
- 6.5.4 This Clause 6.5 shall be without prejudice to the Trustees' powers under Clause 5.5.6 to suspend payments from the Fund and review and if appropriate revise the Payment Percentage.

SECTION SEVEN

7 UK Asbestos Trust Costs

7.1 Introduction

7.1.1 To the extent that they do not constitute Hercules Recovery Costs which can on the Review Date specified in the Hercules Payment Agency Agreement immediately following the time at which they are incurred be recovered under paragraphs 19.8.4 Thirdly or 19.8.4 Fourthly of the CVAs, and subject to paragraph 32.8 of the CVAs, the UK Asbestos Trust Costs shall be paid as follows.

7.2 Non-T&N Fund Costs

7.2.1 Until such time (if any) as the balance of the Remuneration Reserve is transferred to the Trust pursuant to paragraph 32.3 of the CVAs and for so long as personnel or nominees of Kroll Limited comprise the majority of the board of directors of The T&N Asbestos Trustee

Company Limited, the Supervisors shall pay the Non-T&N Fund Costs from the Remuneration Reserve. In circumstances where personnel or nominees of Kroll Limited do not comprise the majority of the board of directors of The T&N Asbestos Trustee Company Limited, the Non-T&N Fund Costs shall be paid out of the relevant Company's Non-T&N Fund in priority to payments to relevant holders of Non-T&N Trust Claims.

7.3 Other UK Asbestos Trust Costs

7.3.1 Subject to paragraph 18.1.2 of the CVAs, all UK Asbestos Trust Costs may, at the Supervisors' sole discretion, be paid from the Remuneration Reserve. If the Supervisors are unable or choose not to pay all or any such costs from the Remuneration Reserve, they shall be paid as follows.

T&N Fund Costs and T&N Hercules Fund Costs

- 7.3.2 The T&N Fund Costs shall be paid out of the T&N Fund, in priority to payments to the holders of T&N Trust Claims.
- 7.3.3 The T&N Hercules Fund Costs shall, subject to paragraphs 19.8.4 Thirdly and 19.8.4 Fourthly of the CVAs, be paid out of the T&N Hercules Fund. If the T&N Hercules Fund does not contain sufficient funds, the balance of such costs shall be paid out of the T&N Fund, and the T&N Fund shall be reimbursed for such costs from any Hercules Recoveries that are subsequently received by the Trust pursuant to paragraph 19.8.4 of the CVAs, which shall constitute T&N Hercules Fund Assets.

Chester Street Fund Costs and Chester Street Hercules Fund Costs

- 7.3.4 The Chester Street Fund Costs shall be paid out of the Chester Street Fund in priority to payments to the relevant holders of Chester Street Trust Claims.
- 7.3.5 The Chester Street Hercules Fund Costs shall, subject to paragraphs 19.8.4 Thirdly and 19.8.4 Fourthly of the CVAs, be paid out of the Chester Street Hercules Fund. If the Chester Street Hercules Fund does not contain sufficient funds, the balance of such costs shall be paid out of the Chester Street Fund, and the Chester Street Fund shall be reimbursed for such costs from any Hercules Recoveries that are subsequently received by the Trust pursuant to paragraph 19.8.4 of the CVAs, which shall constitute Chester Street Hercules Fund Assets.

Costs not attributable to any one particular Fund

7.3.6 The remuneration, costs and expenses (if any) of the Trustees which are incurred in the administration of the Trust and which are not attributable to any one particular Fund shall, to the extent that they are not paid from the Remuneration Reserve, be apportioned between the T&N Fund, the Chester Street Fund and each Non-T&N Fund by reference to the number of

Trust Claims which are asserted or which are projected to be asserted against the Trust and which are attributable to such Funds. The reasonable costs and expenses:

- (a) which are allocated to the T&N Fund shall be met out of the T&N Fund Assets;
- (b) which are allocated to the Chester Street Fund shall be met out of the Chester Street Fund Assets; and
- (c) which are allocated to each Non-T&N Fund shall subject to paragraph 18.1.2 of the CVAs be met out of the relevant Non-T&N Fund Assets.

Hercules Recovery Costs

7.3.7 To the extent that any costs paid pursuant to this Clause properly constitute Hercules Recovery Costs, but are nevertheless paid from the sources referred to in Clauses 7.2 and 7.3, the Trustees shall be entitled to recover an amount equal to such Hercules Recovery Costs pursuant to the application set out in paragraph 19.8.4 Fourthly of the CVAs as if such amount were Hercules Recovery Costs which remained outstanding.

EL Asbestos Insurance Fund Costs

7.3.8 Subject to paragraph 28.2 of the CVAs, the EL Asbestos Insurance Fund Costs shall be paid out of the EL Asbestos Insurance Fund in priority to payments to the holders of EL Asbestos Insured Trust Claims in accordance with and subject to the terms of this Trust Deed and the TDP. If the EL Asbestos Insurance Fund does not contain sufficient funds to permit payment of EL Asbestos Insurance Fund Costs in full, the balance of such costs shall be apportioned between the T&N Fund and each Non-T&N Fund by reference to the number of Trust Claims which are or which are projected to be asserted against the Trust and which are attributable to such funds. The costs and expenses which are allocated to the T&N Fund shall be met out of the T&N Fund Assets and the costs and expenses which are allocated to each Non-T&N Fund shall, subject to paragraph 18.1.2 of the CVAs, be met out of the relevant Non-T&N Fund Assets.

7.4 Remuneration Fund

- 7.4.1 Pursuant to paragraph 32.3 of the CVAs, at the termination of the last CVA, the Supervisors shall pay to the Trust any funds remaining in the Remuneration Reserve to be held on trust and to form the Remuneration Fund. The Trustees shall apply the Remuneration Fund in accordance with paragraphs 18 and 32 of the CVAs, Clause 17 of the Deed, and Clause 7.4.2 of this TDP.
- 7.4.2 After the establishment of the Remuneration Fund, the Trustees may pay the UK Asbestos Trust Costs from the sums standing to the credit of the Remuneration Fund. If it is insufficient to meet such costs, the arrangements set out in Clauses 7.2 and 7.3 shall apply.

SECTION EIGHT

8 Trustees' Powers of Amendment

- 8.1.1 Subject to ensuring that the Core Objective stated in Clause 1.3 of the TDP is maintained, the Trustees may by deed or written resolution amend this TDP and/or the Deed in order to:
 - (a) make such amendments as may from time to time appear to be in the best interests of the beneficiaries of the Trust, including but not limited to amendments necessary to obtain charitable status for all or part of the Trust; or
 - (b) take account of advances in scientific or medical knowledge, or other changes in circumstances or in the law; or
 - (c) comply with any directions or orders of the Court.
- 8.1.2 Notwithstanding the foregoing, no amendment to the TDP or Trust Deed may be made which has or may have the effect of:
 - (a) creating a liability for, increasing the liability of, or reducing the amount receivable by, any company in the Federal-Mogul Group to or from any person, or
 - (b) materially adversely affecting the position of FMC or of any other company in the Federal-Mogul Group,

without the prior written consent of:

- until such time as the Administration Order in respect of T&N is discharged, FMC (such consent not to be unreasonably withheld); or
- (ii) following the discharge of the Administration Order in respect of T&N and if T&N at the relevant time remains a Subsidiary of FMC, T&N (such consent not to be unreasonably withheld); or
- (iii) following the discharge of the Administration Order in respect of T&N and if T&N at the relevant time does not remain a Subsidiary of FMC, FMC (such consent not to be unreasonably withheld).

SECTION NINE

9 Governing law and disputes

- 9.1 This TDP is governed by and shall be interpreted in accordance with the laws of England and Wales.
- 9.2 Any disputes arising out of or in connection with the Trust Deed, this TDP or the operations of the Trust shall be subject to the exclusive jurisdiction of the High Court of Justice of England and Wales, save that where provision is made for an alternative form of dispute resolution in the Trust Deed or this TDP, such alternative form of dispute resolution shall prevail.

Schedule 1

Disease Level	Presumptive Medical and Exposure Criteria	
i. Mesothelioma	(1) Diagnosis by an accredited respiratory specialist ¹ or for Trust Claims in respect of deceased Injured Persons assessed by Expedited Review, a medical certificate issued by the appropriate authority for the relevant jurisdiction certifying the cause of death as mesothelioma.	
	(2) As to employees: Credible evidence of <u>material</u> Asbestos Exposure in the course of employment by T&N or any relevant Company and such Asbestos Exposure commencing at least 10 years before first diagnosis.	
	(3) In other cases: Credible evidence of <u>material</u> Asbestos Exposure caused by the activities of T&N or any relevant company and such Asbestos Exposure commencing at least 10 years before first diagnosis.	
ii. Lung Cancer	 Diagnosis by an accredited respiratory specialist of (a) a primary lung cancer, and (b) asbestosis or credible evidence of Asbestos Exposure sufficient to have caused asbestosis according to the Helsinki Criteria, and 	
	(2) As to employees: Credible evidence of Asbestos Exposure in the course of employment by T&N or any relevant Company for a minimum period of 6 months and such Asbestos Exposure commencing at least 10 years before first diagnosis.	
	(3) In other cases: Credible evidence of Asbestos Exposure caused by the activities of T&N or any relevant Company for a minimum period of 6 months and such Asbestos Exposure commencing at least 10 years before first diagnosis.	
iii. Asbestosis	(1) Diagnosis by an accredited respiratory specialist and assessment of disability as mild, moderate or severe according to the Scale in Schedule 2.	
	(2) As to employees: Credible evidence of Asbestos Exposure in the course of employment by T&N or any relevant Company for a minimum period of 6 months and such Asbestos Exposure commencing at least 15 years before first diagnosis.	
	(3) In other cases: Credible evidence of Asbestos Exposure caused by the activities of T&N or any relevant Company for a minimum period of 6 months and such Asbestos Exposure commencing at least 15 years before first diagnosis.	
iv. Diffuse Pleural Thickening	(1) Diagnosis by an accredited respiratory specialist and assessment of disability as mild, moderate and severe according to the Scale in Schedule 2.	
	(2) As to employees: Credible evidence of Asbestos Exposure in the	

All references to an accredited respiratory specialist include a retired Consultant in respiratory medicine.

1

	Disease Level	Presumptive Medical and Exposure Criteria	
		course of employment by T&N or any relevant Company and suc Asbestos Exposure commencing at least 15 years before firs diagnosis.	
		(3) In other cases: Credible evidence of Asbestos Exposure caused by the activities of T&N or any relevant Company and such Asbestos Exposure commencing at least 15 years before first diagnosis.	
۷.	Pleural Disease Not Causing Disability	(1) Diagnosis by a qualified doctor based on radiological evidence, and	
		(2) As to employees: Credible evidence of material Asbestos Exposure in the course of employment by T&N or any relevant Company and such Asbestos Exposure commencing at least 15 years before first diagnosis.	
		(3) In other cases: Credible evidence of Asbestos Exposure caused by the activities of T&N or any relevant Company and such Asbestos Exposure commencing at least 15 years before first diagnosis.	

Schedule 2

The Disability Rating Scale For Disease Levels III and IV

Disability Percentage	Symptoms	Lung Function Impairment
0%	Not breathless on exercise	None
10%	Breathless on prolonged or heavy exertion	Mild
20%	Breathless on walking uphill or climbing stairs or on hurrying on level ground	Mild
30%	Breathless at normal pace for age walking on level ground	Moderate
40%	Breathless on walking 100 yards or climbing one flight of stairs at a normal pace	Moderate
50%	Breathless on walking 100 yards at a slow pace or climbing one flight of stairs at a slow pace	Moderate
60%	Breathlessness prevents walking 100 yards at a slow pace without stopping or climbing one flight of stairs without stopping	Severe
70%	Breathlessness prevents activity outside the home without assistance or supervision	Severe
80%	Breathlessness limits activities to within the home	Severe
90%	Able to walk only a few steps because of breathlessness	Severe
100%	Bed and chair bound, totally dependent on carers because of breathlessness	Severe

All assessments must be based upon comparison with a healthy person of the same sex and of a similar age.

Schedule 3

1. <u>Trust Claims made under the laws of England, Wales, Northern Ireland and Scotland</u> (save for Trust Claims concerning deceased Injured Persons where Asbestos Disease (Levels I, II or III) was causative of death)

Disease Level	Expedited Review Value	<u>Maximum Value</u>
I Mesothelioma	<u>£134,000</u> £120,000 £100,000	<u>£323,000</u> £288,000 £287,500 £250,000
II Lung Cancer	<u>£112,000</u>	<u>£258,000</u> £230,000 £200,000
III Asbestosis		
Mild	<u>£32,000</u> £29,000 £28,750 £25,000	<u>£65,000</u> £58,000 £57,500 £50,000
Moderate	<u>£65,000</u> £58,000 £57,500 £50,000	<u>£129,000</u> £115,000 £100,000
Severe	<u>£103,000</u> £92,000 £80,000	<u>£258,000</u> £230,000 £200,000
IV Diffuse Pleural Thickening		
Mild	£22,000 £20,000 £17,500	£52,000 £46,000 £40,000
Moderate	£45,000 £40,000 £35,000	£103,000 £92,000 £80,000
Severe	£78,000 £70,000 £60,000	<u>£194,000</u> £173,000 £172,500 £150,000
V Pleural Disease Not Causing Disability	<u>£4,500</u> £4,000 £3,500 £3,000	<u>£52,000</u>

2. <u>Trust Claims concerning deceased Injured Persons which are made under the laws of</u> England, Wales and Northern Ireland where Asbestos Disease (Levels I, II or III) was causative of death

Disease Level	Expedited Review Value	Maximum Value
I Mesothelioma	£155,000 £130,000 £110,000	<u>£375,000</u>
II Lung Cancer	£131,000 £110,000 £90,000	<u>£309,000</u>
III Severe Asbestosis	<u>£124,000</u> £104,000 £103,500 £90,000	<u>£309,000</u> £260,000 £225,000

3. <u>Trust Claims concerning deceased Injured Persons which are made under the laws of</u> <u>Scotland where Asbestos Disease (Levels I, II or III) was causative of death</u>

Disease Level	Expedited Review Value	Maximum Value
I Mesothelioma	<u>£179,000</u> £160,000 £150,000 £140,000	<u>£401,000</u> £358,000 £335,000 £325,000

II Lung Cancer	<u>£143,000</u> £128,000 £120,000 £110,000	<u>£340,000</u>
III Severe Asbestosis	<u>£143,000</u> <u>£128,000</u> <u>£120,000</u> £110,000	<u>£340,000</u>

For the avoidance of doubt, in Schedule 3 Trust Claims "made under the laws of England and Wales, Scotland and Northern Ireland" means Trust Claims where the Asbestos Exposure of the Injured Person occurred, respectively, in England and Wales, Scotland or Northern Ireland.